\documentclass[a4paper,twoside,11pt]{amsart}

\usepackage{amsfonts,amssymb,amscd,amsmath,enumerate}
\usepackage[greek, french, english]{babel}
\usepackage{graphicx}
\usepackage[all]{xy}
\usepackage{enumerate}
\usepackage{color}
\usepackage[T1]{fontenc}



 %\usepackage[active]{srcltx}
 %\usepackage{showkeys}

\newtheorem{theorem}{Theorem}[section]
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}


\newtheorem{claim}{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{notation}[theorem]{Notation}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{example}[theorem]{Example}
\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{Remark}[theorem]{Remarks}
\newtheorem{conjecture}[theorem]{Conjecture}
%-----------------------------------------------------------------------------------------------
\newlength{\szer}
\newcommand{\Teiss}[2]{%
\settowidth{\szer}{$\displaystyle\frac{#1}{#2}$}%
\setlength{\szer}{0.5\szer}%
\left\{\hspace{\szer}%
\raisebox{0.14ex}{\makebox[0pt]{$\displaystyle\frac{#1}{\phantom{#2}}$}}%
\raisebox{-0.14ex}{\makebox[0pt]{$\displaystyle\frac{\phantom{#1}}{#2}$}}%
\hspace{\szer}\right\}%
}
%To write an elementary Newton polygon type \Teiss{a}{b}; a=numerator

\renewcommand{\qedsymbol}{$\square$}
\def \findem{\smallskip\hfill\vrule width 5 pt height 7 pt depth - 2 pt\smallskip}
\def\Par{{\par\noindent}}
\def\A{{\mathbf A}}
\def\Aa{{\mathcal A}}
\def\B{{\mathcal B}}
\def\C{{\mathbf C}}
\def\F{\mathbf F}
\def\Ff{\mathcal F}
\def\J{{\mathcal{J}}}
% \def\L{\mathbf L} %(normally the Polish L)
\def\Ll{{\mathcal L}}
\def\N{{\mathbf N}}
\def\Newton{{\mathcal N}}
\def\O{{\mathcal O}}
\def\P{{\mathbf P}}
\def\Pp{{\mathcal P}}
\def\Pf{{\mathfrak P}}
\def\Kf{{\mathfrak K}}
\def\Q{{\mathbf Q}}
\def\Qq{{\mathcal Q}}
\def\R{\mathbf R}
\def\Rr{{\mathcal R}}
\def\Ss{{\mathcal S}}
\def\T{{\mathbf T}}
\def\Tt{{\mathcal T}}
\def\Z{{\mathbf Z}}
\def\Z{\mathbf Z}

\def\G{{\Gamma}}
\def\a{\alpha}
\def\b{\beta}
\def\g{\gamma}
%\def\d{\delta}
\def\D{\Delta}
\def\f{\varphi}
\def\p{\pi}
\def\h{\psi}
%\def\l{\lambda}
\def\r{\rho}
\def\s{\sigma}
\def\t{\tau}
\def\e{\varepsilon}
\def\og{{\omega}}
\def\z{\zeta}

\def\orb{\mathrm {orb}}
\def\int{\mathrm{int}}
\def\ord{\mathrm {ord}}
\def\codim{\mathrm{codim}}
\def\tr{{\underline{t}}}
\def\y{\wedge}
\def\cprime{$'$}
\providecommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\thinspace}

\makeindex


%\def\Pp{{\Par\medskip\noindent}}
\def\limproj{\mathop{\oalign{lim\cr\hidewidth$\longleftarrow$\hidewidth\cr}}}
\def\limind{\mathop{\oalign{lim\cr\hidewidth$\longrightarrow$\hidewidth\cr}}}
\def\elem(#1,#2){  \{ \frac{#1}{\overline{\ #2\ }}\}  }
\def\dessin#1#2#3{\vbox to #2%w
  {\vfill\special{picture #3}\hrule width#1 height 0pt depth 0pt}}


\begin{document}
\begin{thebibliography}{7}

\bibitem{B-N-S} M.S. Barnab\'e, J. Novacoski, and M. Spivakovsky, On the structure of the graded algebra associated to a valuation,
J. Algebra, \textbf{560} (2020), 667-679.

\bibitem{B}N. Bourbaki, \textit{\'{E}l\'ements de math\'ematique. Alg\`ebre Commutative, Chap. I-VIII}, Masson, Paris

\bibitem{B-G} W. Bruns and J. Gubeladze, Polytopes, rings, and K-Theory, \textit{Springer Monographs in Mathematics}, Springer 2009.

\bibitem{Co} I.S. Cohen, \textit{On the structure and ideal theory of complete local rings}, Transactions of the A.M.S., Vol. 59, No.1 (Jan. 1946), 54-106.

\bibitem{C-G-P} V. Cossart, C. Galindo, and O. Piltant, \textit{Un exemple effectif de gradu\'e non noetherien associ\'e \`a une valuation divisorielle}, Annales Inst. Fourier, \textbf{50}, No. 1 (2000), 105-112.

\bibitem{C} S. D. Cutkosky, On Finite and Nonfinite Generation
of Associated Graded Rings of Abhyankar Valuations, in \textit{G.-M. Greuel, Luis Narv\'aez Macarro, Sebasti\`a Xamb\'o (eds.), Singularities, Algebraic Geometry, Commutative
Algebra, and Related Topics}, Festschrift for Antonio Campillo, Springer Nature 2018. https://doi.org/10.1007/978-3-319-96827-8-20

\bibitem{C2}S. D. Cutkosky, \textit{Local uniformization of Abhyankar valuations}, 
Michigan Mathematical Journal \textbf{71} (4), (2022) 859-891.

\bibitem{C-T} S. D. Cutkosky and B. Teissier, \textit{Semigroups of valuations on Local Rings}, Michigan Math. J., \textbf{57}, (2008), 173-193.

\bibitem{C-M-T} S. D. Cutkosky, H. Mourtada, and B. Teissier, \textit{ On the construction of valuations and generating sequences on hypersurface singularities} Algebraic Geometry, Compositio, 8 (6) (2021) 705-748. doi:10.14231/AG-2021-022

\bibitem{C-V} S. D. Cutkosky and P. A. Vinh, \textit{Valuation semigroups of two dimensional local rings}, Proceedings of the London Math. Soc,  (3) 108 (2014), no. 2, 350-384.  \textit{ArXiv:1105.1448v1.}

\bibitem{dF-T} A. de Felipe and B.Teissier, \textit{Valuations and henselization}, Math. Annalen, 377 (2020) No. 3-4, 935-967. https://doi.org/10.1007/s00208-020-01970-1. February 2020.

\bibitem{E-M} D. Eisenbud and B. Mazur, \textit{Evolutions, symbolic squares, and Fitting ideals}, J. reine u. ang. Math., \textbf{488}, (1997), 189-201.

\bibitem{E-S} D. Eisenbud and B. Sturmfels, \textit{Binomial ideals}, Duke Math. J., \textbf{84}, (1996), 1-45.

\bibitem{G} R. Gilmer, \textit{Commutative semigroup rings}, Chicago Lectures In Mathematics, The University of Chicago Press, 1984.

\bibitem{G-T} R. Goldin and B. Teissier, {\em Resolving plane branch singularities with
one toric morphism}, {\it in} ``Resolution of Singularities, a research textbook in tribute to Oscar Zariski'',
Birkha\"user, Progress in Math. No. 181, 2000, 315-340.

\bibitem{GP-T1}
P.~Gonz\'alez P\'erez and B. Teissier, \textit{Embedded resolutions of non necessarily normal affine toric varieties},
Comptes-rendus Acad. Sci. Paris, Ser.1, \textbf{(334)} (2002), 379-382. Available at http://webusers.imj-prg.fr/~bernard.teissier/

\bibitem{G-L-S} G-M Greuel, C. Lossen, and E. Shustin, Introduction to singularities and deformations, Springer, 2007.

\bibitem{EGA4.1} A. Grothendieck, J. Dieudonn\'e, El\'ements de G\'eom\'etrie Alg\'ebrique, Chap. IV, premi\`ere partie, \textit{Pub. Math. I.H.E.S}, No.20 (1964)

\bibitem{HOST1} J. Herrera, M. A. Olalla, M. Spivakovsky, and B. Teissier,
  \textit{Extending a valuation centered in a local domain to its formal completion}, Proceedings of the London Mathematical Society, (3) 105 (2012) 571-621,

\bibitem{HOST2} J. Herrera, M. A. Olalla, M. Spivakovsky, and B. Teissier,  Extending valuations to formal completions. in \textit{Valuation theory in interaction},  Proceedings of the second international conference on valuation theory, Segovia--El Escorial, 2011. Edited by A. Campillo, F-V. Kuhlmann and B. Teissier. European Math. Soc. Publishing House, Congress Reports Series, Sept. 2014, 252--265.

\bibitem{HOST3} J. Herrera, M. A. Olalla, M. Spivakovsky, and B. Teissier,  Extending valuations of local domains to complete local domains without changing the value group: the case of algebras essentially of finite type over a field. (In preparation)

\bibitem{K-M} Kahle and E. Miller, \textit{Decompositions of commutative monoid congruences and binomial ideals},  Algebra and Number Theory, 8 (2014), 1297-1364.

\bibitem{K} K. Kedlaya, \textit{The algebraic closure of the power series field in positive characteristic}, Proc. Am. Math. Soc., 129, No.12, (2001), 3461-3470 . See the erratum in arXiv:1508.01836: On the algebraicity of generalized power series. 

\bibitem{K-K} H. Knaf,  F.-V. Kuhlmann, \textit{Abhyankar places admit local uniformization in any characteristic}, Ann. Sci. \'Ecole Norm. Sup. \textbf{38}  (4)  (2005),   833--846.

\bibitem{M} M. Moghaddam, \textit{A construction for a class of valuations of the field $k(X_1,\ldots, X_d,Y)$ with large value group}, J. of Algebra, \textbf{319} (2008) 2803-2829.

\bibitem{N} B.H. Neumann, \textit{On ordered division rings}, Trans. Amer. Math. Soc. \textbf{66} (1949), 202-252.

\bibitem{PP} P. Popescu-Pampu, Approximate roots, in \textit{Valuation theory and its applications, Vol. II} (Saskatoon, SK, 1999), 285-321, Fields Inst. Commun., 33, Amer. Math. Soc., Providence, RI, 2003. 

\bibitem{Ro} P. Roquette, History of valuation theory. I, in \textit{Valuation theory and its applications, Vol. I }, (Saskatoon, SK, 1999), 291-355, Fields Inst. Commun., 32, Amer. Math. Soc., Providence, RI, 2002. 

\bibitem{Sp} M. Spivakovsky, \textit{Valuations in function fields of surfaces}, American J. of Math., Vol. 112, No. 1 (Feb., 1990), pp. 107-156.

\bibitem{Te}
B. Teissier, Cycles \'evanescents, sections planes, et conditions de Whitney. 
in \textit{Singularit\'es \`a Carg\`ese (Rencontre Singularit\'es en G\'eom\'etrie Analytique, Institut d'\'etudes Scientifiques de Carg\`ese, 1972)}, pp. 285--362. Asterisque, Nos. 7-8, 1973.

\bibitem{Te1}
B. Teissier, Valuations, deformations, and toric geometry.
In \textit{Valuation Theory and its applications, Vol. II}, Fields Inst. Commun. 33, AMS., Providence, RI., 2003, 361-459.  Available at http://webusers.imj-prg.fr/~bernard.teissier/

\bibitem{Te2} B. Teissier, \textit{A viewpoint on local resolution of singularities}, Oberwolfach Workshop on Singularities, September 2009. Oberwolfach Reports, Vol. 6, No. 3, 2009. European Math. Soc. Publications. Available at http://webusers.imj-prg.fr/~bernard.teissier/

\bibitem{Te3} B. Teissier, Overweight deformations of affine toric varieties and local uniformization, in \textit{Valuation theory in interaction},  Proceedings of the second international conference on valuation theory, Segovia--El Escorial, 2011. Edited by A. Campillo, F-V. Kuhlmann and B. Teissier. European Math. Soc. Publishing House, Congress Reports Series, Sept. 2014, 474-565. 

\bibitem{Te4} B. Teissier, Two points of the boundary of toric geometry, in  \textit{"Singularities, Algebraic Geometry, Commutative Algebra and Related Topics. Festschrift for Antonio Campillo on the occasion of his 65 th Birthday"}, G.-M. Greuel, L. Narv\'aez Macarro and S. Xamb\'o-Descamp, Editors, Springer 2018, 107-117.

\bibitem{Te5} B. Teissier, Some ideas in need of clarification in resolution of singularities and the geometry of discriminants, to appear in \textit{Mathematics Moving Forward}, Springer Lecture Notes in Mathematics no. 2313.

\bibitem{T} M. Temkin, \textit{Inseparable local uniformization}, J. of Algebra, \textbf{373} (2013), 65-119.

\bibitem{V} M. Vaqui\'e, Valuations. In \textit{Resolution of singularities, a research  textbook in tribute to Oscar Zariski}, Progress in Math., No. 181, Birkha\"user (2000), 539--590.

\bibitem{V1} M.~Vaqui\'e, \textit{Extension d'une valuation}, Trans. Amer. Math. Soc., 359, No. 7 (2007), 3439-3481 (electronic).

\bibitem{Z} O. Zariski, \textit{Local Uniformization of algebraic varieties, }Annals of Math., Vol. 41, N0. 4, 1940, 852-896.

\bibitem{Z2} O. Zariski, \textit{The moduli problem for plane branches}, Course given at the Centre de Math\'ematiques de l'Ecole Polytechnique, Paris, October--November 1973, University Lecture Series, 39 (2006). English translation by Ben Lichtin.

\bibitem{Z-S} O. Zariski and P. Samuel, \textit{Commutative Algebra, Vol. II}, Van Nostrand 1960. Reprint: Graduate Texts in Math., Vol. 29, Springer 1975.
\end{thebibliography}

\bibliographystyle{amsplain}
\setcounter{tocdepth}{2}
\title{EQUATIONS FOR FORMAL TORIC DEGENERATIONS }
\author{Bernard Teissier}
\address{Universit\'e Paris Cit\'e and Sorbonne Universit\'e, CNRS, IMJ-PRG, F-75013 Paris, France}

\email{bernard.teissier@imj-prg.fr}

\keywords{Toric geometry, Valuations, toric degeneration, local uniformization}


\pagestyle{myheadings}
\markboth{\rm Bernard Teissier}{\rm Toric degeneration}
\renewcommand\rightmark{B. Teissier}
\renewcommand\leftmark{Equations for formal Toric degenerations}

\subjclass[2000]{14M25, 14E15, 14B05}


\dedicatory{ \textit {}}
\maketitle
\begin{abstract}
Let $R$ be a complete equicharacteristic noetherian local domain and $\nu$ a valuation of its field of fractions whose valuation ring dominates $R$ with trivial residue field extension. The semigroup of values of $\nu$ on $R\setminus \{0\}$ is not finitely generated in general. We produce equations in an appropriate generalized power series ring for the algebra encoding the degeneration of $R$ to the toric graded algebra  ${\rm gr}_\nu R$ associated to the filtration defined by $\nu$. We apply this to represent $\nu$ as the limit of a sequence of Abhyankar semivaluations (valuations on quotients) of $R$ with finitely generated semigroups. \end{abstract}
\maketitle
%\providecommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\thinspace}
\tableofcontents

\begin{section}{Introduction}
For us a valuation is an inclusion of $R$ in a valuation ring $(R_\nu,m_\nu)$ of its field of fractions $K$. If the valuation is centered at $m$, which means that $m_\nu\cap R=m$, the dimension of the valuation is defined to be the transcendence degree of $R_\nu/m_\nu$ over $R/m$. A local uniformization of the valuation $\nu$ on $R$ is a regular local ring $R'$ which is essentially of finite type over $R$ and dominated by $R_\nu$.\par
We continue here the program, presented in \cite{Te1}, \cite{Te3} and \cite{Te5}, to prove local uniformization for valuations on equicharacteristic excellent local domains $(R,m)$ with an algebraically closed residue field. Here is its structure:
\begin{enumerate}
\item Reduce the general case to the case of \textit{rational} valuations, those with trivial residue field extension: $R_\nu/m_\nu=R/m$;
\item Reduce the case of rational valuations to the case where $R$ is complete:
\item Assuming that $R$ is complete, reduce the case of a rational valuation to that of Abhyankar valuations (rational rank=${\rm dim}R$);  
\item Assuming that $R$ is complete, prove the uniformization theorem for Abhyankar valuations using embedded resolution of affine toric varieties, which is blind to the characteristic, applied to the toric graded ring ${\rm gr}_\nu R$ associated to the filtration defined by the valuation.
\end{enumerate}
Step 1) was dealt with in \cite[Section 3.6]{Te1}. Step 2), except in the cases of rank one or Abhyankar valuations, is waiting for the proof of a conjecture about the extension of valuations of $R$ to its completion  (see \cite[*Proposition 5.19*]{Te1}, \cite[Conjecture 1.2]{HOST2} and \cite[Problem B]{Te5}), which is in progress. Step 4) was made in \cite{Te3}. The essential remaining hurdle is step 3).\par The main result of this paper is the production of a sequence of Abhyankar semivaluations $(\nu_{B_t})_{t\in\N}$ of $R$ approximating more and more closely our rational valuation $\nu$ in an embedded manner. Their rational rank is the rational rank of $\nu$. We believe that they have the property that appropriate toric embedded uniformizations as in \cite{Te3} of the $\nu_{B_t}$ close enough to $\nu$ will also uniformize $\nu$. Since we think that the approximation result is of independent interest we make this paper available without waiting for the final proof.
\par\medskip
A valuation $\nu$ with value group $\Phi$ on a subring $R$ of $K$ determines a filtration of $R$ by the ideals $$\Pp_\varphi(R)=\{x\in R/\nu(x)\geq \varphi\}\  {\rm and}\ \  \Pp^+_\varphi (R)=\{x\in R/\nu(x)> \varphi\}\ {\rm for}\ \varphi\in \Phi.$$
We note that this is not in general a filtration in the usual sense because the totally ordered value group $\Phi$ of $\nu$ may not be well ordered so that an element of $\Phi$ may not have a successor. However, the successor ideal  $ \Pp^+_\varphi (R)$ is well defined and if $\varphi$ has a successor $\varphi^+$ we have $\Pp^+_\varphi(R)=\Pp_{\varphi^+}(R)$.\par\noindent The graded ring ${\rm gr}_\nu R=\bigoplus_{\varphi\in \Phi}\Pp_\varphi(R)/\Pp^+_\varphi(R)$ associated to the $\nu$-filtration on $R$ is the graded $k$-subalgebra of ${\rm gr}_\nu K$ whose non zero homogeneous elements have degree in the sub-semigroup\footnote{I follow a tradition of calling semigroup of values of a valuation what is actually a monoid.} $\Gamma=\nu(R\setminus\{0\})$ of $\Phi$. In this text we shall deal with subrings $R$ of the valuation ring $R_\nu$, so that ${\rm gr}_\nu R$ is a graded subalgebra of ${\rm gr}_\nu R_\nu$ since by definition $\Pp_\varphi(R)=\Pp_\varphi(R_\nu)\cap R$, and $\Gamma$ is contained in the non-negative part $\Phi_{\geq 0}$ of $\Phi$.\par In view of the defining properties of valuation rings, for all $\varphi\in\Phi$ any two elements of $\Pp_\varphi(R_\nu)\setminus \Pp^+_\varphi(R_\nu)$ differ by multiplication by a unit of $R_\nu$ so that their images in $\Pp_\varphi(R)/\Pp^+_\varphi(R)$ differ by multiplication by a non zero element of $k_\nu=R_\nu/m_\nu$. Thus, the homogeneous components of ${\rm gr}_\nu R_\nu$ are all one dimensional vector spaces over $k_\nu$.\par \textit{If the valuation $\nu$ is rational}, the same is true of the non zero homogeneous components of ${\rm gr}_\nu R$ as $k=R/m$ vector spaces. Then one has a toric description of ${\rm gr}_\nu R$ as follows:\par\noindent
Since $R$ is noetherian the semigroup $\Gamma$ is well ordered and so has a minimal system of generators $\Gamma=\langle \gamma_1,\ldots ,\gamma_i,\ldots\rangle$, where $\gamma_{i+1}$ is the smallest non zero element of $\Gamma$ which is not in the semigroup $\langle\gamma_1,\ldots, \gamma_i\rangle$. generated by its predecessors. We emphasize here that the $\gamma_i$ are indexed by an ordinal $I\leq \omega^h$, where $h$ is the rank (or height) of the valuation. The graded $k$-algebra ${\rm gr}_\nu R$ is then generated by homogeneous elements $(\overline\xi_i)_{i\in I}$ with ${\rm deg}\overline\xi_i=\gamma_i$ and we have a surjective map of graded $k$-algebras
$$k[(U_i)_{i\in I}]\longrightarrow {\rm gr}_\nu R,\ \ \ U_i\mapsto \overline\xi_i,$$ 
where $k[(U_i)_{i\in I}]$ is the polynomial ring in variables indexed by $I$, graded by giving $U_i$ the degree $\gamma_i$.\par\noindent Its kernel  is generated by binomials $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L},\ \lambda_\ell\in k^*$, where $U^m$ represents a monomial in the $U_i$'s. These binomials correspond to a generating system of relations between the generators $\gamma_i$ of the semigroup, and for all practical purposes we can think of ${\rm gr}_\nu R$ as the semigroup algebra over $k$ of the semigroup $\Gamma$. This is asserted in \cite[Proposition 4.7]{Te1} and made more precise in subsection \ref{gral} below .\textit{We assume that the set $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L}$ is such that none of the binomials is in the ideal generated by the others.} Since the semigroup of weights is well ordered and there are only finitely many binomials of a given weight, this can be achieved by a transfinite cleaning process eliminating at each step the binomials of least weight belonging to the ideal generated by binomials $U^{m^\ell}-\lambda_\ell U^{n^\ell}$, necessarily of smaller weight.\par
An important ingredient in the approach presented in \cite{Te1} and \cite{Te3} and continued here is the fact that by a result of Pedro Gonz\'alez P\'erez and the author (see \cite{Te1}, 6.2 and \cite{GP-T1}), we have embedded resolutions by birational toric maps of affine toric varieties (of finite embedding dimension) such as ${\rm Spec}k[t^\Gamma]$ when the semigroup $\Gamma$ is finitely generated.\par\noindent By general results in \cite[\S 2.3]{Te1} the ring $R$ is a deformation of ${\rm gr}_\nu R$ and the basic observation is that for rational valuations, when ${\rm gr}_\nu R$ is finitely generated, not only is the faithfully flat specialization of $R$ to ${\rm gr}_\nu R$ equidimensional, so that we have equality in Abhyankar's inequality, but the total space of this toric degeneration is "equiresolvable at $\nu$" in the sense that there exist generators $(\xi_i)_{i\in I}$ of the maximal ideal of $R$ whose initial forms $(\overline\xi_i)_{i\in I}$ are generators of  the $k$-algebra ${\rm gr}_\nu R$ and some birational toric maps \textit{in the coordinates $(\xi_i)_{i\in I}$} which provide, in the coordinates $(\overline\xi_i)_{i\in I}$, an embedded resolution of singularities of the affine toric variety ${\rm Spec}{\rm gr}_\nu R$, and when applied to the generators $\xi_i$ of the maximal ideal of $R$ and localized at the point (or center) picked by $\nu$ provide an embedded local uniformization of the valuation $\nu$.\par Of course, in the general case where ${\rm gr}_\nu R$ is not finitely generated even after replacing $R$ by some birational modification along $\nu$, we must adapt this notion of "$\nu$-equiresolvability" since ${\rm Spec}{\rm gr}_\nu R$ does not even have a resolution of singularities, embedded or not.\par\noindent By a theorem of Piltant (see \cite{Te1}, proposition 3.1) we know however that its Krull dimension ${\rm dim}{\rm gr}_\nu R$ is the rational rank $r(\nu)$ of the valuation.\par\noindent For rational valuations of a noetherian local domain Abhyankar's inequality then reads as the following inequality of Krull dimensions:
$${\rm dim}{\rm gr}_\nu R\leq {\rm dim}R.$$ Those for which equality holds, called Abhyankar valuations, were the subject of \cite{Te3}, where it was shown that their graded algebra was quasi-finitely generated\footnote{This means that there is a local $R$-algebra $R'$ essentially of finite type dominated by the valuation ring $R_\nu$ (a birational $\nu$-modification) such that ${\rm gr}_\nu R'$ is finitely generated. For more on finite and non finite generatedness of ${\rm gr}_\nu R$, see \cite{C}.}, and that they could be uniformized in an embedded manner\footnote{A proof of local uniformization for Abhyankar valuations of algebraic function fields, with assumptions on the base field, had appeared in \cite{K-K} and a more general one in \cite{C2}. See also \cite{T}.}.\par The proof reduced the excellent equicharacteristic case to the complete equicharacteristic case using the good behavior of Abhyankar valuations with respect to completion. The case of complete equicharacteristic local domains was then reduced to a combinatorial problem using a valuative version of Cohen's structure theorem to obtain equations describing the specialization of $R$ to ${\rm gr}_\nu R$.\par\noindent
The geometric interpretation of the first part of the valuative Cohen theorem is that after embedding ${\rm Spec}R$ in an affine space $\A^{\vert I\vert}(k)$ by the generators $(\xi_i)_{i\in I}$, the valuation $\nu$ becomes the trace on ${\rm Spec}R$ of a \textit{monomial} valuation (a.k.a. weight in the sense of \cite[Definition 2.1]{Te3}) on the the ambient possibly infinite dimensional affine space.\par\noindent 
%---------------------
The interpretation of the second part is that the image of the embedding in question is defined by a (possibly infinite) set of equations which constitutes an \textit{overweight deformation} (see definition \ref{overweight} below) of a set of binomial equations defining the generalized affine toric variety ${\rm Spec}{\rm gr}_\nu R$. The classical Cohen structure theorem presents a complete equicharacteristic noetherian local ring $R$ as a quotient of a power series ring in finitely many variables by an ideal generated by overweight deformations of the homogeneous polynomials defining the associated graded ring ${\rm gr}_mR$ as a quotient of a polynomial ring.

\begin{example}\label{ex1} Let $R$ be the algebra of an algebroid plane branch over an algebraically closed field $k$. That is, a complete equicharacteristic one dimensional local domain with residue field $k$. The integral closure is $R\subset k[[t]]$ and $R$ has only one valuation $\nu$: the $t$-adic valuation. The set $I=(\gamma_0,\gamma_1,\ldots ,\gamma_g)$ of degrees of a minimal set of generators $(\overline\xi_i)_{i\in I}$ of the $k$-algebra ${\rm gr}_\nu R\subset k[t]$, ordered by weight, is in characteristic zero in order preserving bijection with the set $(\beta_0, \beta_1, \ldots ,\beta_g)$ of characteristic Puiseux exponents. For positive characteristic see \cite[Example 6.2]{Te1} or \cite{Te4}. Representatives $\xi_i\in R$ of the $(\overline\xi_i)_{i\in I}$ embed the curve in $\A^{g+1}(k)$ as a complete intersection defined by overweight deformations of the binomials defining the monomial curve ${\rm Spec}k[t^\Gamma]$. (read \cite[Example 5.69]{Te1} in the formal power series context). \end{example}

When strict inequality holds, which implies that the $k$-algebra ${\rm gr}_\nu R$ is not finitely generated,  $\nu$-equiresolvability should mean that some \textit{partial} toric resolutions of the singularities of the toroidal scheme ${\rm Specgr}_\nu R$, affecting only finitely many of the $(\overline\xi_i)_{i\in I}$, when applied to the elements $\xi_i\in R$ instead of the $\overline\xi_i\in {\rm gr}_\nu R$, should uniformize the valuation $\nu$ on ${\rm Spec}R$.\par The problem is to find the appropriate finite subset of the set $L$ of binomial equations written above.  It should generate a prime ideal in a finite set of variables and have the property that some of the embedded toric resolutions of this prime binomial ideal uniformize the valuation. Theorem \ref{AbhApprox} at the end of this paper at least provides candidates. \par
\end{section}
\begin{section}{Valued complete equicharacteristic noetherian local rings and the valuative Cohen theorem}
\begin{subsection}{The graded algebra of a rational valuation is a twisted semigroup algebra}\label{gral}
Let $\nu$ be a rational valuation centered in a noetherian local domain $R$ with residue field $k$. denoting by $\overline k$ the algebraic closure of $k$, given a minimal system of generators $(\overline\xi_i)_{i\in I}$ of the $k$-algebra ${\rm gr}_\nu R$, a simple algebraic argument based on the fact that the group $\Phi$ has a finite rational rank shows that there exist elements $\rho_i\in (\overline k)^*$ such that the map of $\overline k$-algebras $\overline k\otimes_k{\rm gr}_\nu R\to \overline k[t^\Gamma]$ determined by $1\otimes_k\overline\xi_i \mapsto \rho_i t^{\gamma_i}$ is an isomorphism of graded algebras. By \cite[ Theorems 17.1 and 21.4]{G} we know that the Krull dimension of $k[t^\Gamma]$ is the rational rank $r$ of $\nu$ and by Piltant's Theorem that ${\rm gr}_\nu R$ is also an $r$-dimensional domain. Considering the surjective composed map
$$\overline k[(U_i)_{i\in I}]\longrightarrow \overline k[t^\Gamma],\ \ U_i\mapsto \rho_it^{\gamma_i},$$all that is needed is that the $U_i=\rho_i$ are non zero solutions of the system of binomial equations $(U^{m^\ell}-\lambda_\ell U^{n^\ell}=0)_{\ell\in L}$ which generate the kernel. There exists a finite set $F\subset I$ such that the $(\gamma_i)_{i\in F}$ rationally generate the value group $\Phi$ of our valuation. The binomial equations involving only the variables $(U_i)_{i\in F}$ are finite in number and have non zero solutions $\rho_i\in (\overline k)^*$ by the nullstellensatz and quasi-homogeneity. All the other $\gamma_i$ are rationally dependent on those. The result then follows by a transfinite induction in the well ordered set $I$: assuming the algebraicity to be true for binomial equations corresponding to the relations between the $(\gamma_j)_{j\in F\cup\{j\in I\vert j<i\}}$ and choosing non zero algebraic solutions $\rho_j$, we substitute $\rho_j$ for $U_j$ in the binomials involving the variables $(U_j)_{j\in F\cup\{j\in I\vert j<i\}}$ and $U_i$ to obtain binomial equations in $U_i$ with coefficients in $k((\rho_j)_{j\in F\cup\{j\in I\vert j<i\}})$. By rational dependence we know there is at least one such equation, of the form $U_i^{n_i}-m_i(\rho)=0$ where $m_i(\rho)$ is a non zero Laurent term in the $(\rho_j)_{j\in F}$, and in $k((\rho_j)_{j\in F\cup\{j\in I\vert j<i\}})[U_i,U_i^{-1}]$ the ideal generated by the equations coming from all binomials involving $U_i$ is principal so that up to multiplication by a root of unity we have a well defined solution $\rho_i\in k^*$ and this gives the result. So we have proved the following slight generalization of known results (compare with \cite[Proposition 4.7]{Te1} and, for the case of submonoids of $\Z^r$, \cite[Exercise 4.3]{B-G}):
\begin{proposition}\label{smgrp}Let $\Gamma$ be the semigroup of values of a rational valuation $\nu$ on a noetherian local domain $R$ with residue field $k$. Given a minimal set $(\overline\xi_i)_{i\in I}$ of homogeneous generators of the $k$-algebra ${\rm gr}_\nu R $, with ${\rm deg}\xi_i=\gamma_i$, the map which to an isomorphism $\rho$ of graded $\overline k$-algebras $$\rho\colon \overline k\otimes_k{\rm gr}_\nu R \simeq \overline k[t^\Gamma]$$
associates the family of elements $(\rho (1\otimes_k\overline\xi_i)t^{-\gamma_i})_{i\in I}$ of $(\overline k)^*$ defines a bijection from the set of such isomorphisms to the set of $\overline k$-rational points of  ${\rm Specgr}_\nu R$ all of whose coordinates in ${\rm Spec}\overline k[(U_i)_{i\in I}]$ are non zero; a generalized torus orbit. This set is not empty.
\end{proposition}
The graded algebra ${\rm gr}_\nu R$ is a \textit{twisted semigroup algebra} in the sense of \cite[Definition 10.1]{K-M}. In particular, if $k$ is algebraically closed, there are many isomorphisms of graded $k$-algebras from ${\rm gr}_\nu R$ to $k[t^\Gamma]$.\par\noindent
 For a different viewpoint on the same result, and some interesting special cases, see \cite{B-N-S}. 
\subsection{Overweight deformations of prime binomial ideals (after \cite[Section 3]{Te3})}\label{ow}
Let $w\colon \N^I\rightarrow \Phi_{\geq 0}$ be a semigroup map defining a weight on the variables $(u_i)_{i\in I}$ of a polynomial or power series ring over a field $k$, with values in the positive part $\Phi_{\geq 0}$ of a totally ordered group $\Phi$ of finite rational rank. If we are dealing in finitely many variables, the semigroup of values taken by the weight is well-ordered by a theorem of B. Neumann in \cite[Theorem 3.4]{N}. \par Let us consider the power series case, in finitely many variables, and the ring $S=k[[u_1,\ldots ,u_N]]$. Consider the filtration of $S$ indexed by $\Phi_{\geq 0}$ and determined by the ideals $\Qq_\varphi$ of elements of weight $\geq \varphi$, where the weight of a series is the minimum weight of a monomial appearing in it. Defining similarly $\Qq^+_\varphi$ as the ideal of elements of weight $>\varphi$, the graded ring associated to this filtration is the polynomial ring $$
\bigoplus_{\varphi\in \Phi_{\geq 0}}\Qq_\varphi/\Qq^+_\varphi=k[U_1,\ldots ,U_N],$$
with $U_i={\rm in}_wu_i$, graded by ${\rm deg}U_i=w(u_i)$.

\begin{definition}\label{overweight}
Given a weight $w$ as above, a (finite dimensional) overweight deformation is the datum of a prime binomial ideal $(u^{m^\ell}-\lambda_\ell u^{n^\ell})_{1\leq \ell \leq s},\ \lambda_\ell\in k^*$, of $S=k[[u_1,\ldots ,u_N]]$ such that the vectors $m^\ell-n^\ell\in \Z^N$ generate the lattice of relations between the $\gamma_i=w(u_i)$, and of series 
$$\begin{array}{lr}
F_1=u^{m^1}-\lambda_1 u^{n^1}+\Sigma_{w(p)>w(m^1)} c^{(1)}_pu^p\\
F_2=u^{m^2}-\lambda_2 u^{n^2}+\Sigma_{w(p)>w(m^2)} c^{(2)}_pu^p\\
.....\\
F_\ell=u^{m^\ell}-\lambda_\ell u^{n^\ell}+\Sigma_{w(p)>w(m^\ell)} c^{(\ell)}_pu^p\\
.....\\
F_s=u^{m^s}-\lambda_s u^{n^s}+\Sigma_{w(p)>w(m^s)} c^{(s)}_pu^p\\
\end{array}\leqno{(OD)}
$$ in $k[[u_1,\ldots ,u_N]]$ such that, with respect to the monomial order determined by $w$, they form a standard basis for the ideal which they generate: their initial forms generate the ideal of initial forms of elements of that ideal.\par\noindent 
Here we have written $w(p)$ for $w(u^p)$ and the coefficients $c^{(\ell)}_p$ are in $k$.\par
The dimension of the ring $R=k[[u_1,\ldots ,u_N]]/(F_1,\ldots ,F_s)$ is then equal to the dimension of $k[[u_1,\ldots ,u_N]]/(u^{m^1}-\lambda_1 u^{n^1},\ldots ,u^{m^s}-\lambda_s u^{n^s})$, which is the rational rank of the group generated by the weights of the $u_i$.\par\noindent
The structure of overweight deformation endows the ring $R$ with a rational valuation. The valuation of a nonzero element of $R$ is the maximum weight of its representatives in $k[[u_1,\ldots ,u_N]]$. It is a rational Abhyankar valuation.
\end{definition}
\end{subsection}

\begin{subsection}{A reminder about the valuative Cohen theorem}\label{reminder} When one wishes to compare an equicharacteristic noetherian local domain $R$ endowed with a rational valuation $\nu$ with its associated graded ring ${\rm gr}_\nu R$, the simplest case is when $R$ contains a field of representatives $k$ and the $k$-algebra ${\rm gr}_\nu R$ is finitely generated. After re-embedding ${\rm Spec} R$ in an affine space where ${\rm Spec}{\rm gr}_\nu R$ can be embedded one can again, at least if $R$ is complete, write down equations for the faithfully flat degeneration of ${\rm Spec}R$ to ${\rm Spec}{\rm gr}_\nu R$ within that new space (see \cite[\S 2.3]{Te1}) and use their special form as overweight deformations to prove local uniformization for the valuation $\nu$ as explained in \cite[\S 3]{Te3}. However, without some strong finiteness assumptions on the ring $R$ and the valuation $\nu$, one cannot hope for such a pleasant situation. The valuative Cohen theorem shows that when $R$ is complete one can obtain a similar state of affairs without any other finiteness assumption than the noetherianity of $R$.\par
Let $\nu$ be a rational valuation on a complete equicharacteristic noetherian local domain $R$. Since $R$ is noetherian, the semigroup of values $\Gamma=\nu(R\setminus\{0\})$ is countable and well ordered, and has a minimal set of generators $(\gamma_i)_{i\in I}$ which is indexed by an ordinal $I\leq \omega^{h(\nu)}$ according to \cite[Appendix 3, Proposition 2]{Z-S}. Consider variables $(u_i)_{i\in I}$ in bijection with the $\gamma_i$ and the $k$-vector space of all sums $\Sigma_{e\in E}d_eu^e$ where $E$ is any set of monomials in the $u_i$ and $d_e\in k$.  Since the values semigroup $\Gamma$ is combinatorially finite this vector space is in fact, with the usual multiplication rule, a $k$-algebra $\widehat{k[(u_i)_{i\in I}]}$, which we endow with a weight by giving $u_i$ the weight $\gamma_i$. Combinatorial finiteness means that there are only finitely many monomials with a given weight, and we can enumerate them according to the lexicographic order of exponents. Thus, we can embed the set of monomials $u^m$ in the well ordered lexicographic product $\Gamma\times \N$, where the finite set of monomials with a given weight is lexicographically ordered (see \cite[\S 4]{Te1}). Combinatorial finiteness also implies that the initial form of every series with respect to the weight filtration is a polynomial so that:\par\noindent \textit{ The graded algebra of $\widehat{k[(u_i)_{i\in I}]}$ with respect to the weight filtration is the polynomial algebra $k[(U_i)_{i\in I}]$ with $U_i={\rm in}_wu_i$, graded by giving $U_i$ the weight $\gamma_i$.}\par
The $k$-algebra $\widehat{k[(u_i)_{i\in I}]}$ is endowed with a monomial valuation given by the weight $w(\Sigma_{e\in E}d_eu^e)={\rm min}_{d_e\neq 0}w(u^e) $. This valuation is rational since all the $\gamma_i$ are $>0$. Note that $0$ is the only element with value $\infty$ because here $\infty$ is an element larger than any element of $\Gamma$. With respect to the "ultrametric" given by $u(x,y)=w(y-x)$, the algebra $\widehat{k[(u_i)_{i\in I}]}$ is spherically complete\footnote{A \emph{pseudo-convergent, {\rm or} pseudo-Cauchy} sequence of elements of $\widehat{k[(u_i)_{i\in I}]}$ is a sequence $(y_\tau)_{\tau\in T}$ indexed by a well ordered set $T$ without last element, which satisfies the condition that whenever $\tau<\tau' <\tau"$ we have $w(y_{\tau'}-y_\tau)<w(y_{\tau"}-y_{\tau'})$ and an element $y$ is said to be a \emph{pseudo-limit} of this pseudo-convergent sequence if $w(y_{\tau'}-y_\tau)\leq w(y-y_\tau)$ for $\tau ,\tau'\in T,\ \tau<\tau'$. Spherically complete means that every pseudo-convergent sequence has a pseudo-limit.} by \cite[Theorem 4.2]{Te3}.
%++++++++++++++++
\begin{lemma}\label{appear} Let $A,B$ be two sets of (not necessarily distinct) monomials. If a monomial $u^c$ appears infinitely many times among the monomials $u^{a+b}$ with $u^a\in A,\ u^b\in B$, then at least one decomposition $c=a+b$ must appear infinitely many times. Thus, some $u^a$ or $u^b$ with $c=a+b$ must appear infinitely many times in $A$ or $B$ respectively.
\end{lemma}
\begin{proof} The combinatorial finiteness of the semigroup $\Gamma$ implies that the number of posssible distinct decompositions $c=a+b$ is finite. Thus, at least one of them must occur infinitely many times. \end{proof}
\begin{definition}\label{closure} Let $J$ be an ideal in $\widehat{k[(u_i)_{i\in I}]}$. The closure of $J$ with respect to the ultrametric $w$ is the set of elements of $\widehat{k[(u_i)_{i\in I}]}$ which are a transfinite sum of elements of $J$. That is, the set of elements which can be written $y=\sum_{\tau\in T}(y_{\tau+1}-y_\tau)$ where $T$ is a well ordered set, $(y_\tau)_{\tau\in T}$ is a pseudo-convergent sequence and for each $\tau\in T$ we have $y_{\tau+1}-y_\tau\in J$.
\end{definition}
\begin{proposition}\label{closideal}The closure of an ideal of $\widehat{k[(u_i)_{i\in I}]}$ is an ideal.
\end{proposition}
\begin{proof} This follows from the fact that a transfinite series exists in $\widehat{k[(u_i)_{i\in I}]}$ if any given monomial appears only finitely many times. Since that must be the case for the sum $y$ of the definition, it is also the case for the sum or difference of two such elements, and for the product with an element of the ring. All the terms of such sums are in $J$.

\end{proof}
\begin{proposition}\label{h(u)} For any series $h(u)\in\widehat{k[(u_i)_{i\in I}]}$ which is without constant term, the series $\sum_{a=0}^\infty h(u)^a$ converges in $\widehat{k[(u_i)_{i\in I}]}$.
\end{proposition}
\begin{proof}Since there is no constant term, we have to prove that any non constant monomial can appear only finitely many times in the series. If the value group is of rank one, the result is clear by the archimedian property. We proceed by induction on the rank. If $\Psi_1$ is the largest non trivial convex subgroup of the value group $\Phi$, we can write $h(u)=h_1(u)+h_2(u)$ where $h_1(u)$ contains all those terms in $h(u)$ which involve only variables with weight in $\Psi_1$. Then $h(u)^a=\sum_{i=0}^a{{a}\choose{i}}h_1(u)^ih_2(u)^{a-i}$. If a monomial appears infinitely many times in the $(h(u)^a)_{a\in \N}$ then applying lemma \ref{appear} to the sets of monomials appearing in the $(h_1(u)^i)_{i\in \N}$ and $(h_2(u)^j)_{j\in \N}$ respectively shows that some monomial must appear infinitely many times in the collection of series $(h_1(u)^i)_{i\in \N}$, which we exclude by the induction hypothesis applied to $\Psi_1$, or some monomial must appear infinitely many times in the collection of series  $(h_2(u)^j)_{j\in \N}$, which is impossible because the semigroup $\Phi_+\setminus \Psi_1$ is archimedian since $\Phi/\Psi_1$ is.
\end{proof}
\begin{corollary}\label{local} The ring $\widehat{k[(u_i)_{i\in I}]}$ is local, with maximal ideal $\hat m$ consisting of series without constant term and residue field $k$. The maximal ideal is the closure of the ideal generated by the $(u_i)_{i\in I}$.
\end{corollary}
Indeed the proposition shows that the non invertible elements of the ring are exactly those without constant term, which form an ideal. The second assertion follows from the fact that any series without constant term is a transfinite sum of $w$-isobaric polynomials in the $u_i$.
\begin{proposition}\label{hensel}The local ring $\widehat{k[(u_i)_{i\in I}]}$ is henselian.
\end{proposition}
\begin{proof} This follows from the fact that it is spherically complete, that the value group of $w$ is of finite rational rank, and the characterization of henselian rings by the convergence of pseudo-convergent sequences of \'etale type found in \cite[Proposition 3.7]{dF-T}.
\end{proof}
\begin{corollary}\label{algebraic} The henselization of the localization $k[(u_i)_{i\in I}]_{\hat m\cap k[(u_i)_{i\in I}]}$ of the polynomial ring $k[(u_i)_{i\in I}]$ at the maximal ideal which is the ideal of polynomials without constant term, is contained in $\widehat{k[(u_i)_{i\in I}]}$. 
\end{corollary}
\begin{proof} By proposition \ref{h(u)}, any element of $k[(u_i)_{i\in I}]$ which is not in the maximal ideal is invertible in $\widehat{k[(u_i)_{i\in I}]}$, which is henselian.
\end{proof}
\begin{corollary}\label{substitute} If the series $h(u)$ has no constant term, given any other series $g(u)\in \widehat{k[(u_i)_{i\in I}]}$ and a monomial $u^m$, writing $g(u)=\sum_{a\in \N}g_a(u)(u^m)^a$, where no term of a $g_a(u)$ is divisible by $u^m$, the substitution $\sum_{a\in \N}g_a(u)h(u)^a$ gives an element of $\widehat{k[(u_i)_{i\in I}]}$: in this ring, we can substitute a series without constant term for any monomial, and in particular for any variable.\end{corollary}
\begin{proof} Since any monomial appears finitely many times in $g(u)$, by combinatorial finiteness, any monomial can appear only finitely many times among the series $(g_a(u))_{a\in\N}$.\par\noindent Should a monomial appear infinitely many times among the terms of the sum $\sum_{a\in \N}g_a(u)h(u)^a$, by lemma \ref{appear} applied to the set of monomials appearing in the $(g_a(u))_{a\in\N}$ and the set of monomials appearing in the $(h(u)^a)_{a\in\N\setminus\{0\}}$, a monomial would have to appear infinitely many times among the $h(u)^a$ and this would contradict proposition \ref{h(u)}.
\end{proof}
\textit{In conclusion, $\widehat{k[(u_i)_{i\in I}]}$ is regular in any reasonable sense and has almost all of the properties of the usual power series rings except of course noetherianity if $\Gamma$ is not finitely generated. If $\Gamma$ is finitely generated, $\widehat{k[(u_i)_{i\in I}]}$ is the usual power series ring in finitely many variables, equipped with a weight.}
 \begin{theorem}\label{cohen}{{\rm (The valuative Cohen theorem, part 1)}} Let $R$ be a complete equicharacteristic noetherian local domain and let $\nu$ be a rational valuation of $R$. We fix a field of representatives $k\subset R$ of the residue field $R/m$ and a minimal system of homogeneous generators $(\overline\xi_i)_{i\in I}$ of the graded $k$-algebra ${\rm gr}_\nu R$. There exist choices of representatives $\xi_i\in R$ of the $\overline\xi_i$ such that the application $u_i\mapsto \xi_i$ determines a surjective map of $k$-algebras 
 $$\pi\colon \widehat{k[(u_i)_{i\in I}]}\longrightarrow R$$
 which is continuous with respect to the topologies associated to the filtrations by weight and by valuation respectively. The associated graded map with respect to these filtrations is the map
 $${\rm gr}_w\pi\colon k[(U_i)_{i\in I}]\longrightarrow {\rm gr}_\nu R,\ \ U_i\mapsto\overline\xi_i$$
whose kernel is a prime ideal generated by binomials $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L},\ \lambda_\ell\in k^*$.\par\noindent
If the semigroup $\Gamma=\nu(R\setminus\{0\})$ is finitely generated or if the valuation $\nu$ is of rank one, one may take any system of representatives $(\xi_i)_{i\in I}$. \end{theorem}
This is proved in \cite[\S 4]{Te3} and we recall that in view of \cite[Proposition 3.6]{Te3}, \textit{the initial form ${\rm in}_wh(u)$ of an element $h(u)\in \widehat{k[(u_i)_{i\in I}]}$ is in the kernel of ${\rm gr}_w\pi$ if and only if $w(h(u))<\nu (\pi(h(u)))$}.
\begin{remark}\label{HS}1) After giving this result its name, I realized that Cohen's structure theorem had roots in valuation theory. It is an analogue for complete noetherian local rings of a structure theorem for complete valued fields due to H. Hasse and F.K. Schmidt. See \cite[Section 4.3]{Ro}. However, this result was itself inspired by a conjecture of Krull in commutative algebra which directly inspired I.S. Cohen; see \cite[Introduction]{Co}.\par\medskip\noindent
2) The sum $\sigma=\sum u^m$ of \textit{all} monomials is an element of $\widehat{k[(u_i)_{i\in I}]}$ and can be deemed to represent the transfinite product $\prod_{i\in I}(\frac{1}{1-u_i})$ since in the expansion of $\frac{1}{1-u_i}$ each power of $u_i$ appears exactly once and with coefficient $1$.\par\noindent If we consider the canonical map of topological $k$-algebras $\widehat{k[(u_i)_{i\in I}]}\to k[[t^\Gamma]]$ determined by $u_i\mapsto t^{\gamma_i}$ defined in \cite[introduction to section 4]{Te3}, by construction a monomial of weight $\gamma$ in the variables $u_i$ encodes a representation of $\gamma$ as a combination of the generators $\gamma_i$ so that the image of the series $\sigma$ is the generalized generating series $\sum_{\gamma\in\Gamma} p(\gamma) t^\gamma\in k[[t^\Gamma]]$ where $\gamma\mapsto p(\gamma)\in\N$ is a partition function: the number of distinct ways of writing $\gamma$ as a combination with coefficients in $\N$ of the generators $\gamma_i$. We see that it satisfies a generalized version of Euler's identity for the generating series of partitions: $$\sum_{\gamma\in\Gamma} p(\gamma) t^\gamma =\prod_{i\in I}(\frac{1}{1-t^{\gamma_i}}).$$\par\noindent  We observe that the semigroup $\Phi_{\geq 0}$ is itself well ordered and combinatorially finite if and only if its approximation sequence 
$$\N^r_0\subset \N^r_1\subset \cdots \subset \N^r_h\subset \N^r_{h+1}\subset\cdots\Phi_{\geq 0}$$by nested free subsemigoups $\N^r$, where $r$ is the rational rank of $\Phi$,  following from \cite[Proposition 2.3]{Te3}, is finite. Indeed, if it is infinite there must be infinite decreasing sequences of elements of $\Phi_{\geq 0}$ because if the linear inclusion $\N_h^r\subset \N_{h+1}^r\subset \Phi_{\geq 0}$, encoded by a matrix with entries in $\N$, is not an equality, there has to be a basis element of $\N_{h+1}^r$ which is strictly smaller than the minimum of elements of $\N_h^r$. Conversely, there can be no infinite decreasing sequence of elements of $\N^r$ with a monomial ordering where all elements are $\geq 0$.\par Therefore, if $\Phi_{\geq 0}$ is well ordered, we have $\Phi_{\geq 0}=\N^r$ where $r$ is the rational rank of $\Phi$. The same construction as above can be applied when taking variables $u_i$ indexed by all the elements of $\N^r$ instead of only generators and taking $r=1$ we recover the usual partition function $p(n)$ and the usual Euler identity.\par\noindent 
The series $\sum_{\gamma\in\Gamma} p(\gamma) t^\gamma $ can also be viewed as the generalized Hilbert-Poincar\'e series of the graded algebra ${\rm gr}_w\widehat{k[(u_i)_{i\in I}]}$ since $p(\gamma)$ is the dimension of the $k$-vector space generated by the monomials of degree $\gamma$. We note that the Hilbert-Poincar\'e series of ${\rm gr}_\nu k[[t^\Gamma]]=k[t^\Gamma]$ is again, up to a change in the meaning of the letter $t$, the sum $\sum_{\gamma\in\Gamma}t^\gamma$ of all monomials.
\end{remark}
\end{subsection}
%______________________________
\begin{subsection}{The valuative Chevalley Theorem}
In this section we dig a little deeper into the relationship between the valuative Cohen theorem and Chevalley's theorem (see \cite{Te1}, section 5, and \cite{B}, Chap. IV, \S 2, No. 5, Cor.4), which is essential in its proof. Let $R$ be a complete equicharacteristic noetherian local domain and let $\nu$ be a rational valuation centered in $R$. Let us take as set $(\gamma_i)_{i\in I}$ the entire semigroup $\Gamma$. If the valuation $\nu$ is of rank one, this set is of ordinal $\omega$ (see \cite[Appendix 3, Proposition 2]{Z-S}) and cofinal in $\Phi_{\geq 0}$ since by \cite[Theorem 3.2 and \S 4]{C-T} the semigroup $\Gamma$ has no accumulation point in $\R$ because we assume that $R$ is noetherian. Since we have $\bigcap_{\gamma\in \Gamma}\Pp_{\gamma}=(0)$, it is an immediate consequence of Chevalley's theorem that given a sequence of elements of $R$ of strictly increasing valuations, their $m$-adic orders tend to infinity. This is no longer true for valuations of rank $>1$ as evidenced by the following:
\begin{example}\label{nochev} Let $R=k[[x,y]]$ equipped with the monomial rank two valuation $\mu$ with value group $(\Z^2)_{lex}$ such that $\mu(x)=(1,0)$ and $\mu(y)=(0,1)$. The elements $(x+y^i)_{i\geq 1}$ have strictly increasing values $(0,i)$ but their $m$-adic order remains equal to $1$. \end{example}
This is the reason why in the valuative Cohen theorem the representatives of generators of the associated graded algebra have to be chosen. In this section we present some consequences of this choice.\par\medskip
Let $(\gamma_i)_{i\in I}$ be a well ordered set contained, with the induced order, in the non negative semigroup of a totally ordered group $\Phi$ of finite rational rank. By the theorem of B. Neumann in \cite{N} which we have already quoted, the semigroup $\Gamma$ generated by those elements is well ordered. Let
$$(0)=\Psi_h\subset \Psi_{h-1}\subset\cdots\subset\Psi_1\subset\Psi_0=\Phi$$
be the sequence of convex subgroups of $\Phi$, where $h$ is the rank of $\Phi$ and $\Psi_k$ is of rank $h-k$. Let us consider the canonical map $\lambda\colon \Phi\rightarrow \Phi/\Psi_{h-1}$. 
With these notations we can state the following inductive definition:
\begin{definition}\label{definit} A subset $B$ of a well ordered subset $(\gamma_i)_{i\in I}$ of $\Phi_{\geq 0}$ is said to be \textit{initial in $(\gamma_i)_{i\in I}$} when
 \begin{itemize} 
\item If $h=1$, then $B$ is of the form $\{ \gamma_i\vert i\leq i_0\}$.
\item If $h>1$, then $\lambda(B)$ is initial in the set $(\lambda(\gamma_i))_{i\in I}\subset  \Phi/\Psi_{h-1}$ of the images of the $\gamma_i$ and for every $\varphi_{h-1}\in \lambda(B)$ we have that the set of differences $B\cap\lambda^{-1}(\varphi_{h-1})-\tilde\varphi_{h-1}$ is initial in the set of differences $\{\gamma_i)_{i\in I}\}\cap\lambda^{-1}(\varphi_{h-1})-\tilde\varphi_{h-1}\subset (\Psi_{h-1})_{\geq 0}$, where $\tilde\varphi_{h-1}$ is the smallest $\gamma_i$ contained in $\lambda^{-1}(\varphi_{h-1})$.
\end{itemize}
\end{definition}
 One verifies by induction on the rank that the intersection of initial subsets is initial, so that we have an initial closure of a subset $C$ of $(\gamma_i)_{i\in I}$, the intersection of the initial subsets containing $C$.
 \begin{remark}\label{plusone} An initial subset $B$ has the property that if $\gamma_i\notin B$, then $\gamma_{i+1}\notin B$, where as usual $i+1$ is the successor of $i$ in $I$. The converse is false when the rank of the valuation is $>1$ and there are elements without predecessor in $(\gamma_i)_{i\in I}$. It is not true either, if the valuation is of rank $>1$, that if $\gamma_i\in B$ and $\gamma_j<\gamma_i$, then $\gamma_j\in B$ or that if $\gamma_j\notin B$ and $\gamma_k>\gamma_j$, then $\gamma_k\notin B$.\par The useful features of initial sets are what comes now and especially the valuative Chevalley Theorem below which shows that finite initial subsets provide a way of approximating the countable ordinal $I$ indexing the generators of $\Gamma$ by nested finite subsets which is appropriate for the valuative Cohen Theorem.
 \end{remark}
 
 \textit{To simplify notations, we shall sometimes identify each $\gamma_i$ with its index $i$, as in the proof below.}
\begin{proposition}\label{initial}Let the $(\gamma_i)_{i\in I}$ be as above. The initial closure of a finite subset of $(\gamma_i)_{i\in I}$ is finite.
\end{proposition}
\begin{proof} This is a variant of the proof of lemma 5.58 of \cite{Te1}. Let $C$ be a finite subset of $I$. If the valuation is of rank one the ordinal of $I$ is at most $\omega$ (see \cite[Appendix 3, Proposition 2]{Z-S}) and the initial closure of $C$ is the set of elements of $I$ which are less than or equal to the largest element of $C$, and it is finite. Assume now that the result is true for valuations of rank $\leq h-1$ and let $\nu$ be a valuation of rank $h$. Let $\lambda\colon \Phi\to\Phi_{h-1}=\Phi/\Psi_{h-1}$ be the map of groups
corresponding to the valuation $\nu_{h-1}$ of height $h-1$ with which $\nu$ is composed.  Set $C_1=\lambda (C)$ and let $I_1$ index the distinct $(\lambda(\gamma_i))_{i\in I}$. We have a natural monotone map $I\to I_1$, which we still denote by $\lambda$. By the induction hypothesis, we have a finite set $\tilde C_1$ containing $C_1$ and initial in  $I_1$. Define $\tilde C$ as follows: it is the union of finite subsets $\tilde C_{i_1}$ of the $\lambda^{-1}(i_1)$ for $i_1\in \tilde C_1$. Define $\tilde C_{i_1}$ to be the set of elements of $\lambda^{-1}(i_1)$ which are smaller than or equal to the largest element of $C\cap\lambda^{-1}(i_1)$. By \textit{loc. cit.},
Lemma 4, or Proposition 3.17 of \cite{Te1}, each of these sets if finite and $\tilde C=\bigcup_{i_1\in \tilde C_1}\tilde C_{i_1}$ is the initial closure of $C$. \end{proof}
\begin{corollary}\label{approx} Given a finite set $B_0\subset I$, there is a countable sequence $(B_t)_{t\in \N_{>0}}$ of nested finite initial subsets containing $B_0$ whose union is $I$:
$$B_0\subset B_1\subset \cdots\subset B_t\subset B_{t+1}\subset \cdots\subset I$$
\end{corollary}
\begin{proof} Consider the sequence of morphisms
$$\Phi\stackrel{\lambda_{h-1}}\longrightarrow \Phi/\Psi_{h-1}\stackrel{\lambda_{h-2}}\longrightarrow \Phi/\Psi_{h-2}\cdots\cdots\stackrel{\lambda_{2}}\longrightarrow \Phi/\Psi_2\stackrel{\lambda_{1}}\longrightarrow \Phi/\Psi_1\longrightarrow 0.$$
Denote by $\mu_i$ the compositum $\lambda_i\circ\lambda_{i-1}\circ\cdots\circ\lambda_{h-1}\colon\Phi\to\Phi/\Psi_i.$ Let $B_{0,1}\subset \Phi/\Psi_1$ be the initial closure of the union of the image of $B_0$ in $\Phi/\Psi_1$ and the smallest element of the image of $I$ in $\Phi/\Psi_1$ which is not in that image. For each $\varphi_1\in B_{01}$ take the initial closure in $\lambda_{1}^{-1}(\varphi_1)$ of the union of \break$\lambda_{1}^{-1}(\varphi_1)\cap\mu_2(B_0)$ and the smallest element of $\lambda_{1}^{-1}(\varphi_1)\cap\mu_2(I)$ which is not in $\lambda_{1}^{-1}(\varphi_1)\cap\mu_2(B_0)$. If $\varphi_1\notin\mu_1(B_0)$ take the smallest element of $\lambda_{1}^{-1}(\varphi_1)\cap\mu_2(I)$. The union of the results of this construction is a finite initial subset $B_{02}$ of $\mu_2(I)$ which contains $\mu_2(B_0)$. \par\noindent
We repeat the same construction starting from $B_{02}\subset\mu_2(I)$ to build a finite initial subset $B_{03}$ of $\mu_3(I)$, and so on until we have created a finite initial subset $B_1$ of $I$ which strictly contains $B_0$ unless $B_0=I$. Then we apply the same construction replacing $B_0$ by $B_1$ to obtain $B_2$, and so on.\par
  Let us now prove that the union of the $B_t$ is $I$. We keep the notations of Definition \ref{definit} and Proposition \ref{initial}, where $\lambda_{h-1}=\lambda$. The result is true if $\Phi$ is of rank one because a strictly increasing sequence of elements of $I$ is cofinal in $I$ (see  \cite[Theorem 3.2]{C-T}). Let us assume that it is true for groups of rank $\leq h-1$ and assume that the result is not true for rank $h$. Let $\iota$ be the smallest element of $I$ which is not in $\bigcup_{t\in\N} B_t$. Using our induction hypothesis, define $t_0$ as the least $t$ such that $\lambda(\iota)\in \lambda (B_t)$. Let $\tilde\varphi_\iota$ be the smallest element of $I\cap\lambda^{-1}(\lambda(\iota))$. By the definition of initial closure, for $t\geq t_0$ the elements of each set of differences $B_t\cap\lambda^{-1}(\lambda(\iota))-\tilde\varphi_\iota$ are initial in $I\cap \lambda^{-1}(\lambda(\iota))-\tilde\varphi_\iota\subset \Psi_{h-1}$. We are reduced to the rank one case and since by construction the $B_t\cap\lambda^{-1}(\lambda(\iota))$ grow with $t$ we obtain that for large enough $t$ we have $\iota\in B_t\cap\lambda^{-1}(\lambda(\iota))$ and a contradiction.
\end{proof}
\begin{theorem}\label{Chevalley}{{\rm (The valuative Chevalley theorem)}} Let $R$ be a complete equicharacteristic noetherian local domain and let $\nu$ be a rational valuation of $R$. Denote by $\Gamma$ the semigroup of values of $\nu$ and by $(\gamma_i)_{i\in I}$ a minimal set of generators of $\Gamma$. Assume that the set $I$ is infinite and let $D$ be an integer. Let $B_0$ be a finite subset of $I$. There exist elements $(\xi_i)_{i\in I}$ in $R$ such that $\nu(\xi_i)=\gamma_i$ and a finite initial subset $C(D)$ of $(\gamma_i)_{i\in I}$ containing $B_0$ and such that whenever $\gamma_i\notin C(D)$, then $\xi_i\in m^{D+1}$. 
 \end{theorem}
\begin{proof}This is essentially a consequence of the valuative Cohen theorem of \cite{Te3}, or rather of its proof. Recall that the choice of the $\xi_i$ in the proof of the valuative Cohen theorem is such that if the image of $\xi_i$ in a quotient $R/p_q$ is in a power of the maximal ideal, then $\xi_i$ lies in the same power of the maximal ideal in $R$, for all primes $p_q$ which are the centers of valuations with which $\nu$ is composed. So if $p_1$ is the center of the valuation $\nu_1$ of height one with which $\nu$ is composed, we may assume by induction that the result is true for the $\xi_i$ whose value is in the convex subgroup $\Psi_1$ of $\Phi$ associated to $\nu_1$. By a result of Zariski (see \cite[corollary 5.9]{Te1}), we know that the $\nu_1$-adic topology on $R$ is finer than the $m$-adic topology, which means that there exists a $\varphi_1\in\Phi_1=\Phi/\Psi_1$ such that if $\nu_1(\xi_i)\geq \varphi_1$ then $\xi_i\in m^{D+1}$. So we are interested only in those $\xi_i$ whose $\nu_1$-value is positive and less that $\varphi_1$. If $\nu_1(m)>0$, they are finite in number (see \cite[proposition 3.17]{Te1}) so it suffices to add them to the finite set constructed at the previous inductive stage. If not, since $\Phi_1$ is of rank one, there are only finitely many values of the $\nu_1(\xi_i)$ which are positive and less than $\varphi_1$. Let  $\eta$ be one of these. Denote by $\lambda\colon\Phi\to\Phi_1$ the natural projection. Again using the argument of the proof of the valuative Cohen theorem, and in particular Chevalley's theorem applied to the filtration of $\Pp_\eta/\Pp^+_\eta$ by the $\Pp_\varphi/\Pp^+_\eta$ with $\lambda(\varphi)=\eta$, we have that for each $\eta$ there is a $\varphi_0(\eta)\in \lambda^{-1}(\eta)$ such that if $\nu(\xi_i)\in\lambda^{-1}(\eta)$ and  $\nu(\xi_i)\geq\varphi_0(\eta)$ then $\xi_i\in m^{D+1}$. Again by (\cite{Te1}, proposition 3.17) there are finitely many generators of $\Gamma$ in $\lambda^{-1}(\eta)$ which are $\leq \varphi_0(\eta)$. It suffices to add to our finite set the union of these finite sets over the finitely many values $\eta$ that are $\leq \varphi_1$. 
 \end{proof}
 \begin{proposition} If the rational valuation $\nu$ is of rank one, Theorem \ref{Chevalley} is equivalent to the existence for each integer $D>0$ of a finite initial set $C(D)$ in $\Gamma$ such that for $\gamma\notin C(D)$ we have $\Pp_\gamma\subset m^{D+1}$.
 \end{proposition}
 \begin{proof} If the set $I$ is infinite, the $\gamma_i$ are cofinal in $\Phi_{\geq 0}$ since by \cite[Theorem 3.2 and \S 4]{C-T} the semigroup $\Gamma$ has no accumulation point in $\R$ because we assume that $R$ is noetherian. Therefore we have $\bigcap_{i\in I}\Pp_{\gamma_i}=(0)$ where the ordinal of $I$ is $\omega$. By Chevalley's theorem (see \cite[Chap. IV, \S 2, No. 5, Cor. 4]{B}), all but finitely many of the $\xi_i$ are in $m^{D+1}$. Conversely, assuming the result of Theorem \ref{Chevalley}, recall that $\Pp_\gamma$ is generated by the monomials $\xi^e$ of value $\geq \gamma$ and let $s\in I$ be the smallest index such that $t\geq s$ implies that $\xi_t\in m^{D+1}$. Let $(\gamma_i)_{i\in J}$ be the finite set of those $\gamma_j<\gamma_s$. For any monomial $\xi^e$ such that some decomposition of $\nu(\xi^e)$ as sum of $\gamma_i$ contains a $\gamma_t$ with $t\geq s$ we have $\xi^e\in m^{D+1}$. Let $C(D)$ be the set of elements of $\Gamma$ which are $\leq D\gamma_s$. If $\nu(\xi^e)=\gamma >D\gamma_s$, either its decomposition along the $\gamma_i$ contains a $\gamma_t$ with $t\geq s$ or it can be written $\gamma=\sum_{i\in J}a_i\gamma_i$ with $a_i\in \N$ and then since $(\sum_{i\in J}a_i){\rm sup}(\gamma_i)_{i\in J}> D\gamma_s$ we must have $(\sum_{i\in J}a_i)\geq D+1$ and thus $\xi^e\in m^{D+1}$ since the $\xi_i$ are in $m$.\par\noindent
 If the set $I$ is finite, taking as initial set $C(D)$ the set of elements of $\Gamma$ which are $\leq D\gamma_s$, where now $\gamma_s$ is the largest of the $\gamma_i$, the last argument of the proof shows that $\Pp_\gamma\subset m^{D+1}$ for $\gamma\notin C(D)$. \end{proof}
 \begin{remark} 
So we see that in the rank one case Theorem \ref{Chevalley} is indeed equivalent to the result given by Chevalley's theorem.  \end{remark}
 \begin{corollary}\label{chev} In the situation of the valuative Cohen theorem, given an infinite collection of monomials $(u^{q_\ell})_{\ell\in L }$ such that each monomial appears only finitely many times, for any integer $D$ all but finitely many of the $\xi^{q_\ell}$ are in $m^{D+1}$. 
\end{corollary}
\begin{proof} Set $C=C(D)$ and let us consider those monomials $u^{q_\ell}$ which involve only the finitely many variables $u_i$ with $i\in C$. If there are finitely many such monomials, we are done. Otherwise, since each monomial appears only finitely many times, and we are now dealing with finitely many variables, there are only finitely many of those monomials which are such that $\vert q_\ell\vert\leq D$ and we are done again since the $\xi_i\in m$ and by construction all the other monomials are in $m^{D+1}$.
\end{proof}
\begin{remark}Since each monomial appears only finitely many times, the sum $\sum_{\ell\in L}u^{q_\ell}$ exists in $\widehat{k[(u_i)_{i\in I}]}$ and therefore its image $\sum_{\ell\in L}\xi^{q_\ell}$ must converge in $R$, albeit in a transfinite sense. The corollary shows an aspect of this.  \end{remark}
\end{subsection}
%____________________________
\begin{subsection}{Finiteness in the valuative Cohen theorem}\label{finiteness} When the rational valuation $\nu$ on the noetherian complete local domain $R$ is of rank $>1$ and the semigroup $\Gamma$ is not finitely generated, not only does one have to choose the representatives $\xi_i\in R$ of generators of the $k$-algebra ${\rm gr}_\nu R$ carefully (see \cite{Te3}) in order to avoid adding infinitely many times the same element in sums such as $\sum_{i\in I}\xi_i$, but the equations $F_\ell$ whose initial forms are the binomials $(u^{m^\ell}-\lambda_\ell u^{n^\ell})_{\ell\in L},\ \lambda_\ell\in k^*$ also have to be chosen carefully in order to avoid similar accidents when writing elements of the closure in $\widehat{k[(u_i)_{i\in I}]}$ of the ideal which they generate. In this subsection we show how the noetherianity of the local domain $R$ can be used to prove the existence of good choices of equations.
Since the binomials $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L},\ \lambda_\ell\in k^*$ generate the $w$-initial ideal of the kernel $F$ of the map $\pi\colon \widehat{k[(u_i)_{i\in I}]}\longrightarrow R$, for each $\ell\in L$ there is at least one element of the form $u^{m^\ell}-\lambda_\ell u^{n^\ell}+\sum_{w(p)>w(u^{m^\ell})}c_pu^p$ which is in $F$ (overweight deformation of its initial binomial). Let us call such elements $F_\ell$.
%____________________
\begin{proposition}\label{finite} The elements $F_\ell$ can be chosen, without modifying the initial binomial, in such a way that each one involves only a finite number of variables and no monomial $u^p$ appears in infinitely many of them or in infinitely many products $A_\ell F_\ell$, where the $A_\ell$ are terms such that there are at most finitely many products $A_\ell F_\ell$ of any given weight.\par\noindent
 \end{proposition}
\begin{proof} According to equation $(E)$ of subsection \ref{reminder}, the image in $R$ of a topological generator is of the form $\xi^{m^\ell}-\lambda_\ell\xi^{n^\ell}- \sum_pc^{(\ell)}_p\xi^p$ with $\nu(\xi^p)>\nu(\xi^m)$ for all exponents $p$ and of course it is zero in $R$. Since the ring $R$ is noetherian, the ideal generated by the $\xi^p$ which appear in the series is finitely generated, say by $\xi^{e_1},\ldots ,\xi^{e_s}$. Let us choose finitely many of the $\xi_j$ which generate the maximal ideal of $R$, call them collectively $\Xi$ and call $U$ the collection of the corresponding variables in $\widehat{k[(u_i)_{i\in I}]}$ . Then our series can be rewritten as $\xi^{m^\ell}-\lambda_\ell\xi^{n^\ell}- \sum_{j=1}^s G^{(\ell)}_j(\Xi)\xi^{e_j}$, where $ G^{(\ell)}_j(\Xi)\in R$ is a series of monomials in the elements of $\Xi$. Our series is the image of the element \break$u^{m^\ell}-\lambda_\ell u^{n^\ell}- \sum_{j=1}^sG^{(\ell)}_j(U)u^{e_j}$ of $F$.  This element has the same initial binomial as our original series since $w(u^{e_j})>w(u^{m^\ell})$ for all $j$ and involves only finitely many variables. It can replace our original series as element of the ideal $F$ with initial form $u^{m^\ell}-\lambda_\ell u^{n^\ell}$.\par
To prove the second assertion, remember from \cite[\S 4]{Te3} that there is a well-ordering on the set of all monomials obtained by embedding it into $(\N\times\Gamma)_{lex}$ thanks to the finiteness of the fibers of the map $u^e\mapsto w(u^e)$. Fix a choice of $(F_\ell)_{\ell\in L}$ and consider all $k$-linear combinations of the $F_\ell$ of the form $F_\ell -\mu F_{\ell'}$ which do not modify the initial binomial. If the set of monomials $u^e$ which appear infinitely many times as terms in all such linear combinations is not empty, it has a smallest element $u^{e_0}$. Note that we may assume that such a monomial does not appear in the initial binomials because there are only finitely many initial binomials involving a given finite set of variables.\par\noindent Denote by $L_0$ the set of indices $\ell\in L$ such that $u^{e_0}$ appears in $F_\ell$ as above, and  by $L_1\subset L_0$ the finite set of indices $\ell\in L_0$ such that the weight of the initial form of $F_\ell$ is minimal. Denote by $L_2$ the set of indices in $L_0\setminus L_1$ such that the weight of the initial form of $F_\ell$ is minimal, and define recursively in the same manner finite subsets $L_\iota$ indexed by an ordinal.  Now for each $\ell\in L_1$ we can replace $F_\ell$ by $F_\ell-\mu_{\ell,\ell'}F_{\ell'}$ with an $\ell'\in L_2$ without changing the initial binomial, where the constant $\mu_{\ell,\ell'}$ is chosen to eliminate the monomial $u^{e_0}$ from the difference. In so doing we create a new system of overweight deformations of the collection of binomial ideals and cannot add a monomial of lesser weight than $u^{e_0}$ which might appear infinitely many times since $u^{e_0}$ was the smallest in the original family $(F_\ell)_{\ell \in L}$. We continue this operation with $\ell\in L_\iota,\ell'\in L_{\iota+1}$. In the end we have a new system of overweight deformations obtained by linear combinations and in which $u^{e_0}$ cannot appear. This contradiction shows that we can choose the $F_\ell$ so that no monomial appears infinitely many times. \par\noindent It follows from lemma \ref{appear} that under our assumption, for any collection of isobaric polynomials $A_\ell$, the $A_\ell (F_\ell -(u^{m^\ell}-\lambda_\ell u^{n^\ell}))$, cannot contain infinitely many times the same monomial. On the other hand since each $A_\ell$ is an isobaric polynomial, a monomial can appear only finitely many times in all the $A_\ell u^{m^\ell}$ or $A_\ell u^{n^\ell}$, so that altogether no monomial can appear infinitely many times in the collection of series $A_\ell F_\ell$.
\end{proof}
\begin{theorem}\label{gener}{{\rm (The valuative Cohen theorem, part 2)}}
The kernel $F$ of $\pi$ is the closure of the ideal generated by the elements $F_\ell$ obtained as in Proposition \ref{finite} as $U^{m^\ell}-\lambda_\ell U^{n^\ell}$ runs through a set of generators of the kernel of ${\rm gr}_w\pi$. In a slightly generalized sense, these generators form a standard basis of the kernel of $\pi$ with respect to the weight $w$.
\end{theorem}
\begin{proof} Let $h$ be a non zero element of $F$; its weight is finite but its image in $R$ has infinite value. Therefore, ${\rm in}_wh$ belongs to the ideal $F_0$ generated by the $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L}$.\par\noindent Thus, there exists a finite set of elements $(F_\ell)_{\ell\in L_1}$, with $ L_1\subset L$, and isobaric polynomials $(A_\ell)_{\ell\in L_1}$ in the $u_i$ such that $w(h-\sum_{\ell \in L_1}A^{(1)}_\ell F_\ell)>w(h)$. Since this difference is still in $F$, we can iterate the process and build a series indexed by the set $T$ of weights of the successive elements, say $y_\tau$, with $y_0=h, y_1=h-\sum_{\ell \in L_1}A^{(1)}_\ell F_\ell$ and, if $\tau+1$ is the successor of $\tau$ in the index set $T$, $y_\tau-y_{\tau+1}=\sum_{\ell \in L_\tau}A^{(\tau)}_\ell F_\ell$, such that $w(y_{\tau+1})>w(y_\tau)$, so that $w(y_\tau)= w(y_{\tau+1}-y_\tau)$. The sequence $(y_\tau)_{\tau\in T}$  is a pseudo-convergent sequence with respect to the ultrametric $u(x,y)=w(y-x)$ which, according to \cite[ \S 4]{Te3} has $0$ as a pseudo-limit in the spherically complete ring $ \widehat{k[(u_i)_{i\in I}]}$. This is of course not sufficient to prove what we want. However,  in view of proposition \ref{finite}, no monomial can appear in infinitely many of the $y_\tau-y_{\tau+1}$. If $h\neq \sum_{\tau<\rho}(y_{\tau+1}-y_\tau)$, the initial form of $h- \sum_{\tau<\rho}(y_{\tau+1}-y_\tau)$ is in $F_0$ and we can continue the approximation. Thus, the transfinite sum $\sum_{\tau\in T}(y_\tau-y_{\tau+1})$ exists in $\widehat{k[(u_i)_{i\in I}]}$ and, by definition of $T$, has to be equal to $h$ so that $0$ is indeed the limit of the sequence $y_\tau$. This shows that $h$ is in the closure of the ideal generated by the $F_\ell$. \end{proof}
\begin{remark}\label{error}The statement about the $F_\ell$ topologically generating the kernel of $\pi$ is part a) of the asterisked\footnote{An asterisked propositions in that text means it is endowed with hope but not with a proof.}  proposition 5.53 of \cite{Te1}. Part b) of that proposition is incorrect, as the next section shows.
\end{remark}
\textit{From now on we shall assume not only that the $(\xi_i)_{i\in I}$ are representatives in $R$ of the $\overline\xi_i\in {\rm gr}_\nu R$ which make the valuative Cohen theorem valid, but also that each of the equations $F_\ell$ we consider involves only finitely many variables and the $F_\ell$'s topologically generate the kernel of $\pi$}.\par\medskip\noindent
 In view of their definition the $(\xi_i)_{i\in I}$ generate the maximal ideal of $R$. Since $R$ is noetherian there is a finite subset $J\subset I$ such that the elements $(\xi_i)_{i\in J}$ minimally generate this maximal ideal. According to (\cite{Te1}, 5.5), for each $\ell\in I\setminus  J$ there must be among the topological generators $(E)$ of the ideal $F$ one which contains $u_\ell$ as a term. We detail the argument here:\par\noindent The element $\xi_\ell$ is expressible as a series $h((\xi_i)_{i\in J})$. Therefore the series $u_\ell-h((u_i)_{i\in J})$ must belong to the ideal $F$.  There must be an element $H$ of the closure of the ideal generated by the series $(u^{m^\ell}-\lambda_\ell u^{n^\ell}+\sum_{w(p)>w(m^\ell)}c_p^{(\ell)}u^p)_{\ell\in L}$ such that $w(u_\ell-h((u_i)_{i\in J})-H)>\gamma_\ell$. Thus, the series $H$ must contain $u_\ell$ as a term so that at least one of the topological generators of $F$ must contain $u_\ell$ as a term. An argument given in (\cite{Te3}, proof of proposition 7.9) suggests that:
\begin{proposition}\label{lin} Up to a change of the representatives $\xi_i$ we may assume that each variable $u_i$ with $i\in I\setminus J$ appears as a term in a topological generator of ${\rm ker}\pi$ of the form:
$$u^{m^i}-\lambda_i u^{n^i}-g_i((u_j)_{j<i})-u_i,\eqno{(F_i)}$$ where every term of the series $g_i((u_j)_{j<i})$ has weight $>w(u^{m^i})$ and $<w(u_i)$.
\end{proposition}
\begin{proof} Up to multiplication of variables by a constant and corresponding modification of the $\lambda_\ell$, we may assume that for each $i\in I\setminus J$, the variable $u_i$ appears with coefficient minus one in $F_i$. Let $i_1$ be the smallest element of $I\setminus J$ such that there exists an equation $F_i$ as above containing $u_i$ linearly and not having the form described above. We write $g((u_j)_{j<i_1})$ for the series of terms of weight $<i_1$. The remaining part of $F_{i_1}$ can be written $-u_{i_1}+\sum_{w(p)>\gamma_{i_1}}c^{(i_1)}_pu^p$. If we replace the representative $\xi_{i_1}\in R$ by $\xi_{i_1}-\sum_{w(p)>\gamma_{i_1}}c^{(i_1)}_p\xi^p$ without changing the previous $\xi_j$, we modify the equation $F_{i_1}$ into the desired form. We note that in view of the definition of $i_1$ we may assume that no variable of index $<i_1$ appears as a term in $(F_{i_1})$. We then continue with the smallest index $i_2$ such that the corresponding equation does not have the desired form, and obtain the result by transfinite induction.
\end{proof}

\begin{proposition}\label{irr} Each $F_\ell$ is irreducible in $\widehat{k[(u_i)_{i\in I}]}$.
\end{proposition}
\begin{proof} Should it be reducible, since the kernel $F$ of $\pi$ is prime, one of its factors should be in $F$ and by Theorem \ref{gener} and the fact that the kernel of ${\rm gr}_w\pi$ is prime, the initial binomial of $F_\ell$ should be in the ideal generated by the initial forms of the other generators of $F$, which is impossible in view of the assumption we made on the generating binomials.\par
\end{proof}
\begin{remark}
Since the initial forms of the $F_\ell$ are the binomials defining our toric variety ${\rm Spec}{\rm gr}_\nu R$, Theorem \ref{gener} indeed provides equations for the degeneration to ${\rm Spec}{\rm gr}_\nu R$. However, since the binomials $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L}$ come in bulk, it is difficult to make use of them. The next two subsections address this difficulty.
\end{remark}
\end{subsection}

\begin{subsection}{The valuative Cohen theorem for power series rings}\label{cohenreg}
In this section we prove that, in the case where $R$ is a power series ring, the equations provided by the valuative Cohen theorem can then be given a more specific form.\par\noindent
Assume that $R$ is a power series ring and we have chosen representatives $(\xi_i)_{i\in I}$ with which we can apply the valuative Cohen theorem. Since the $\xi_i$ generate the maximal ideal, we can find a subset $J\subset I$ such that the $(\xi_i)_{i\in J}$ form a minimal system of generators of the maximal ideal of $R$. Note that we state nothing about the rational independence of their valuations. 
By Proposition \ref{lin} for each $i\in I\setminus J$ there is a series of the form 
$$u^{m^i}-\lambda_i u^{n^i}-g_i((u_j)_{j<i})-u_i,\eqno{(F_i)}$$
among the topological generators of ${\rm ker}\pi$.\par\noindent Generators of the relations between the values of the $(\xi_i)_{i\in J}$ are encoded by finitely many binomials $(U^{m_\ell}-\lambda_\ell U^{n_\ell})_{\ell \in L_0}$ which according to Theorem \ref{gener} are the initial forms for the weight $w$ of series $(F_\ell)_{\ell \in L_0}$ which are among the topological generators of ${\rm ker}\pi$. 
%__________
\begin{theorem}\label{regular} If $R$ is regular, i.e., a power series ring with coefficients in $k$, the kernel of the map $\pi\colon \widehat{k[(u_i)_{i\in I}]}\to R$ asssociated to a rational valuation $\nu$ by Theorems \ref{cohen} and \ref{gener} is the closure of the ideal generated by the $(F_i)_{i\in I\setminus J}$.
\end{theorem}
\begin{proof} It suffices to show that all the $(F_\ell)_{\ell \in L}$ of Theorem \ref{gener} are in the ideal generated by the $F_i$. It is impossible for any $F_\ell$ to use only variables with indices in $J$ because if that were the case its image in $R$ by $\pi$ would be a non trivial relation between the $(\xi_i)_{i\in J}$, contradicting the fact that these are a minimal set of generators of the maximal ideal in a regular local ring.\par\noindent Given $F_\ell$ with $\ell\in L$, let $i$ be the largest index which is not in $J$ of a variable appearing in $F_\ell$. Let us substitute $F_i+u_i$ in place of $u_i$ in $F_\ell$, according to corollary \ref{substitute}.  Let us denote by $\tilde F_\ell$ the result of this substitution. It involves only finitely many variables, and those whose indices are not in $J$ are all of index $<i$. The difference $F_\ell -\tilde F_\ell$ is a multiple of $F_i$. If $\tilde F_\ell=0$, since $F_\ell$ is irreducible it should be equal to $F_i$ up to multiplication by a unit, which proves the assertion in this case. Thus we may assume that $\tilde F_\ell\neq 0$. Since the initial form ${\rm in}_w\tilde F_\ell$ can have only finitely many factors, the same is true for $\tilde F_\ell$ and since the kernel of $\pi$ is prime, at least one of these factors belongs to it. The variables appearing in this factor are a subset of those appearing in $\tilde F_\ell$. Now we can again take the variable of highest index not in $J$, say $k$, appearing in this factor, make the substitution using the corresponding $F_k$ and repeat the argument; if the result of the substitution is zero, then $F_\ell$ belonged to the ideal generated by $F_i$ and $F_k$. Otherwise we have an irreducible element in ${\rm ker}\pi$ involving only variables of index $<k$ and variables with index in $J$. Since the indices are elements of a well ordered set, after finitely many such steps, either we obtain zero which shows what we want, or we obtain a non zero element of ${\rm ker}\pi$ involving only the variables $(u_i)_{i\in J}$. Again the image of this element by $\pi$ would be a non trivial relation between the elements of a minimal system of generators of the maximal ideal of $R$, which is impossible since $R$ is regular. Thus we obtain a contradiction if $F_\ell$ did not belong to the ideal generated by the $F_i$. 
\end{proof}
\begin{remark}\label{choice} 

1) By the faithful flatness of the toric degeneration, each binomial $u^{m^\ell}-\lambda_\ell u^{n^\ell}$ is the initial binomial of an equation $F_\ell$ and thus the $F_\ell$ are a (generalized) standard basis for the kernel of $\pi$ (see \cite[Proposition 5.53]{Te1}). We do not know whether the $F_i$, which topologically generate the same ideal, are such a standard basis.\par\noindent 
2) Theorem \ref{regular} can be interpreted as a manifestation of the "abyssal phenomenon" of \cite[section 5.6]{Te1}. If the values of the generators $(\xi_i)_{i\in J}$ of the maximal ideal of our power series ring are rationally independent, the semigroup $\Gamma$ is finitely generated, and we have an Abhyankar, even monomial, valuation. If such is not the case, the rational rank of the valuation is strictly less than the dimension of $R$ and we are in the non-Abhyankar case. The semigroup $\Gamma$ is not finitely generated and the binomial relations in ${\rm gr}_\nu R$ expressing the rational dependance of values of the $(\xi_i)_{i\in J}$ cannot give rise to relations in $R$, as they should in view of the faithful flatness of the degeneration of $R$ to ${\rm gr}_\nu R$, because $R$ is regular. The role of the equations $(F_i=0)_{i\in I\setminus J}$ is to prevent their expression in $R$, which would decrease the dimension and introduce singularities, by sending this expression to infinity. Of course one could also point out that something like this is necessary to deform a non-noetherian ring such as ${\rm gr}_\nu R$ into the noetherian $R$, but that leaves out the precise structure of the deformation, which is significant even when ${\rm gr}_\nu R$ is noetherian; see Example \ref{ex1}. \par\noindent
This indicates that it is probably not possible to approximate directly our non-Abhyankar valuation by Abhyankar valuations in this context. However, as we shall see in the sequel, using appropriate truncations of the system of equations $F_\ell$, one can produce a sequence of Abhyankar semivaluations on $R$ (Abhyankar valuations on quotients of $R$) which approximate our valuation $\nu$.\par\noindent
 \end{remark}
\end{subsection}
\begin{subsection}{Lifting the valuation to a power series rings}\label{CohenRegular}
We can present our complete equicharacteristic local domain $R$ as a quotient of a power series ring $S=k[[x_1,\ldots ,x_n]]$ over $k$ by a prime ideal $P$. Our rational valuation $\nu$ then appears as a valuation on $S/P$ which we can compose with the $PS_P$-adic valuation $\mu_1$ of the regular local ring $S_P$ to obtain a rational valuation $\mu$ on $S$. By definition of the symbolic powers $P^{(e)}=P^eS_P\cap S$, the associated graded ring of $S$ with respect to $\mu_1$ is ${\rm gr}_{\mu_1}S=\bigoplus_{e\in \N}\frac{P^{(e)}}{P^{(e+1)}}$.\par
The graded ring ${\rm gr}_\mu S$ is then the graded ring associated to the filtration of ${\rm gr}_{\mu_1}S$  induced on each homogeneous component by the valuation ideals $\Pp_\varphi/P^{(e+1)}$ with $P^{(e+1)}\subseteq \Pp_\varphi\subseteq P^{(e)}$, where $\varphi\in \{e\}\oplus\Phi$.\par  Let us pick a minimal system of generators $p_1,\ldots, p_s$ for the ideal $P$. We note that a regular system of parameters for the regular local ring $S_P$ consists of $n-d$ of the $p_q$'s. We shall assume in the sequel that the dimension of $S$ is minimal, which means that its dimension is the embedding dimension of $R$.
\begin{proposition}\label{compose} 1) The valuation $\mu$ on $S$ obtained as explained above is rational and its group is $(\Z\oplus\Phi)_{lex}$.  \par\noindent 2) The generators $p_1,\ldots ,p_s$ of the ideal $P$ can be chosen so that $\mu(p_1)<\mu(p_2)<\ldots <\mu(p_s)$ and their $\mu$-initial forms ${\rm in}_\mu p_1,\ldots ,{\rm in}_\mu p_s$ are part of a minimal system of generators of ${\rm gr}_\mu S$. All the generators of the semigroup of $\mu$ which are of $\mu_1$-value $>0$ and are not the $\mu (p_q)$ are of value $>\mu (p_s)$.\par\noindent 
\par\noindent 3)
With such a choice the ideal $PS_P$ is generated by $p_1,\ldots ,p_{n-d}$ and in particular  the $\mu_1$-initial forms ${\rm in}_{\mu_1}p_1,\ldots,{\rm in}_{\mu_1}p_{n-d}$ of $p_1,\ldots ,p_{n-d}$ are non zero in $P/P^{(2)}$. \end{proposition}
\begin{proof} Let $S_\mu\subset S_{\mu_1}$ be the inclusion of valuation rings corresponding to the fact that $\mu$ is composed with $\mu_1$ and let $m_{\mu_1}$ be the intersection with $S_\mu$ of the maximal ideal of $S_{\mu_1}$. The valuation ring $R_\nu$ is equal to the quotient $S_\mu/m_{\mu_1}$ and has the same residue field as $S_\mu$.  By (\cite{V}, \S 4), the value group of $\mu$ is an ordered extension of $\Z$ by $\Phi$, which can only be $(\Z\oplus\Phi)_{lex}$.\par If $\mu(p_1)=\mu(p_2)$ for example, their initial forms in ${\rm gr}_\mu S$ are proportional since $\mu$ is rational, and we may replace $p_2$ by $p_2-\lambda p_1$ whose value is $>\mu (p_1)$ if $\lambda$ is chosen appropriately in $k$, and so on. 
In view of the example given on page 200 of \cite{E-M}, some of the series $p_q$ may belong to $P^{(2)}$. However, that is not possible for those which constitute a regular system of parameters for $PS_P$. \footnote{When the field $k$ is of characteristic zero, it is likely that none of the $p_q$ may belong to $P^{(2)}$. We do not know whether $P^{(2)}\subset (x_1,\ldots ,x_n)P$ (see \cite{E-M}) which would give the result immediately by Nakayama's Lemma, but we can obtain a partial result:\par\noindent If $p_q\in P^{(2)}$, there is an $h\notin P$ such that $hp_q\in P^2$. In characteristic zero this implies that $\frac{\partial p_q}{\partial x_i}\in P$ for $i=1,\ldots ,n$. By \cite[Chap. 0, 0.5]{Te}, $p_q$ is then integral over $mP$ so that if we denote by $P_1\subset P$ the ideal generated by those generators of $P$ which are not in $P^{(2)}$, we have $P_1+\overline{mP}=P$. Since the ideal $P$ is integrally closed, by the integral Nakayama Lemma of \cite[lemme 2.4]{Te}, with ${\mathfrak n}_1=0$, we have $\overline{P_1}=P$.}
\par To prove the second assertion of 2), let $\zeta\in S$ be a representative of the smallest element of the semigroup of $\mu$ on $S$ whose $\mu_1$-value is $>0$. This element has to be part of a minimal system of generators of ${\rm gr}_\mu S$. By construction we have $\mu(\zeta)\leq \mu(p_1)$. We also have  ${\rm in}_{\mu_1}\zeta \in P/P^{(2)}$, which is generated as $S/P$-module by the images of the $p_q$, so that $\mu (\zeta)\geq{\rm min}(\mu (p_q))=\mu(p_1)$ and finally $\mu(\zeta)=\mu(p_1)$.
We now proceed by induction and assume that the $\mu$-initial forms of $p_1,\ldots ,p_j$ are the first $j$ generators of ${\rm gr}_\mu S$ whose $\mu_1$-value is $>0$. Since $S$ is complete, we may assume that ${\rm in}_{\mu}p_{j+1}$ is not the initial form of an element of the ideal of $S$ generated by $p_1,\ldots ,p_j$; if it were, we might replace $p_{j+1}$ by $p_{j+1}-\sum_{q=1}^ja_qp_q$ without affecting its role as generator of $P$ while increasing the $\mu$-value by an appropriate choice of the $a_q$, and continue until we either reach a contradiction with the minimality of the system of generators or obtain a generator with the desired property. This is essentially the same as the argument in the proof of lemma 4.6 of \cite{Te3}. Therefore the $\mu$-initial form of $p_{j+1}$ must involve a generator of the graded algebra of degree greater than $\mu (p_j)$, say $\overline\zeta_\ell$, so that ${\rm in}_\mu p_{j+1}=\lambda\overline\xi^\alpha\overline\zeta^e_\ell,\ \lambda\in k^*$, with $e\in \N$ and $\overline\xi^\alpha$ is a monomial in generators of ${\rm gr}_\mu S$ different from $\zeta_\ell$. Unless we have ${\rm in}_\mu p_{j+1}=\lambda\overline\zeta_\ell$, this implies that for a representative $\zeta_\ell\in P$ of $\overline\zeta_\ell$, we have $\mu(\zeta_\ell)<\mu (p_{j+1})$. On the other hand we can write $\zeta_\ell=\sum_{q=1}^ja_qp_q+\sum_{q=j+1}^sa_qp_q$ and the inequality we have just seen implies that the $\mu$-initial form of $\zeta_\ell$ must come from the first sum. But then the $\mu$-initial form of $p_{j+1}$ is the initial form of an element of the ideal of $S$ generated by $p_1,\ldots ,p_j$, and we have a contradiction. This proves that the $\mu$-initial form of $p_{j+1}$ is part of a minimal system of generators of ${\rm gr}_\mu S$. On the other hand, the smallest generator of the semigroup of $\mu$ which is not in the semigroup $\langle \{0\}\oplus \Gamma,{\rm in}_\mu p_1,\ldots ,{\rm in}_\mu p_j\rangle$ must, by the same argument as above, have a value $\geq\mu(p_{j+1})$, so that it is $\zeta_\ell$. Assertion 2) now follows by induction.\par\noindent To prove 3) we notice that among the elements $p_q$, those which minimally generate $PS_P$ cannot be in $P^{(2)}$, and that is the case for $p_1,\ldots p_{n-d}$ as we defined them because of the inequalities on valuations.
   \end{proof}
    \begin{lemma}\label{upgen} Given a finite subset $(\xi_i)_{i\in J}$ of a set $(\xi_i)_{i\in I}$ of representatives of the generators $(\overline\xi_i)_{i\in I}$ of the graded algebra ${\rm gr}_\nu R$, chosen in such a way that the valuative Cohen theorem is valid for them, and where $J$ is such that the $(\xi_i)_{i\in J}$ generate the maximal ideal $m_R$ of $R$, let us denote by $(\eta_i)_{i\in J}$ representatives in $S$ of the $(\xi_i)_{i\in J}$. The $(\eta_i)_{i\in J}$ generate the maximal ideal $m_S$ of $S$.
 \end{lemma}
 \begin{proof} Let $M$ denote the ideal of $S$ generated by the $(\eta_i)_{i\in J}$, $m_R$ and $m_S$ the maximal ideals of $R$ and $S$ respectively. By construction me have $M+P=m_S$ since $S/M+P=R/m_R=k$. But since the dimension of the regular ring $S$ is the embedding dimension of $R$, we must have the inclusion $P\subset m_S^2$. By Nakayama's lemma this implies $M=m_S$.
 \end{proof}
Given the rational valuation $\mu$ of rational rank $r+1< n$ on the power series ring $S=k[[x_1,\ldots, x_n]]$ which we have just built, the generators $\gamma_i$ of the semigroup $\Gamma$ are part of the minimal system of generators of the semigroup $\Delta$ of $\mu$. The other elements of the minimal system of generators of $\Delta$ are the $\mu(p_1),\ldots ,\mu(p_s)\in \Z\oplus\Phi$ by proposition \ref{compose}, and the other members of this minimal system of generators of $\Delta$ which are $>\mu(p_s)$  and which we denote by $\delta_a$.\par\noindent
We therefore have a surjection
$$k[(U_i)_{i\in I}, V_1,\ldots ,V_s, (W_a)_{a\in A}]\rightarrow {\rm gr}_\mu S, \ U_i\mapsto {\rm in}_\mu\eta_i, V_q\mapsto {\rm in}_\mu p_q, W_a\mapsto {\rm in}_\mu\zeta_a.$$
 By the valuative Cohen theorem we can lift the $\xi_i\in R$ to elements $\eta_i\in S$ and choose representatives $\zeta_a$ in $S$ for the elements of ${\rm gr}_\mu S$ of degree $\delta_a$ in such a way that the theorem applies, the $p_q$ being finite in number can be kept, and so we obtain a continuous surjective map of topological $k$-algebra $$\Pi\colon \widehat{k[(u_i)_{i\in I},v_1,\ldots ,v_s,(w_a)_{a\in A}]}\rightarrow S,\ u_i\mapsto \eta_i,\ v_q\mapsto p_q,\ w_a\mapsto \zeta_a,$$
to which we can apply Theorem \ref{regular} and obtain the following key
\begin{lemma}\label{goodinit} Let $\theta\in S$ be an element whose initial form in ${\rm gr}_\mu S$ is part \break of a minimal system of generators of this $k$-algebra and let \break$z\in ((u_i)_{i\in I},(v_q)_{1\leq q\leq s},(w_a)_{a\in A})$ be the corresponding variable. A topological generator $F_z$ of the kernel of $\Pi$ which contains $-z$ as a term according to Theorem \ref{regular} can be chosen so that the following holds: If $\theta$ is one of the $\eta_i$ or one of the $p_q$, the initial binomial of $F_z$ involves only the variables $(u_i)_{i\in I}$. If $\theta$ is one of the $\zeta_a$, each term of the initial binomial of $F_z$ involves some of the $v_j$.
\end{lemma} 
\begin{proof} If $\theta$ is one of the $\eta_i$ the result is clear, since the variables in the initial binomial must be of value less than that of $\eta_i$. If $\theta$ is one of the $p_q$, say $p_\ell$, there cannot even exist a $F_z$ whose initial binomial is not in $k[(u_i)_{i\in I}]$ because if that was the case, both terms of the initial binomial would contain a monomial in the $p_q$ and then $p_\ell$ would be in the ideal generated by the $p_q$ of smaller value, which contradicts the minimality of the system of generators. Now if $w_a$ corresponds to $\zeta_a\in P$ which is not one of the $p_q$, there exist $a_q\in \widehat{k[(u_i)_{i\in I},v_1,\ldots ,v_s,(w_a)_{a\in A}]}$ such that $w_a-\sum_{q=1}^sa_qv_q$ is in ${\rm ker}\Pi$ and is not zero. The process of expressing this difference as a combination 
$$w_a-\sum_{q=1}^sa_qv_q=\sum A_\ell F_\ell$$ of the topological generators of ${\rm ker}\Pi$ begins with expressing the initial form for the weight of that difference as a combination of binomials which are part of a generating system for ${\rm ker}{\rm gr}\Pi$. In a minimal such expression there can be no term which is a monomial in the $U_i$ because there is no such term in the initial form of the difference. Since the values can only increase from there and in view of the form of the valuation $\mu$, the sum $\sum A_\ell F_\ell$ can contain no term using only the $u_i$. On the other hand, since $w_a$ is a term in the difference it must appear as a term in that sum, and we know that one of the $F_\ell$ must contain $w_a$ (or $-w_a$) as a term. It follow that one of the $F_\ell$ appearing in the sum, which by construction has an initial binomial which uses some of the $v_q, w_b$, contains $w_a$ (or $-w_a$)  as a term. 
\end{proof}
Keeping the notations of this section, let us now denote by $\Pp$ the closure of the ideal of $\widehat{k[(u_i)_{i\in I},v_1,\ldots ,v_s,(w_a)_{a\in A}]}$ generated by the $(v_q)_{q=1,\ldots ,s}$ and the $(w_a)_{a\in A}$. We have a commutative diagram:
 \[\xymatrix { 
       \widehat{k[(u_i)_{i\in I},v_1,\ldots ,v_s,(w_a)_{a\in A}]}\ar[d]  \ar[r]^{\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \  \Pi}  &S \ar[d] \\
               \widehat{k[(u_i)_{i\in I}]} \ar[r]^{\pi}&  R }\leqno{(*)}\]
               where the left vertical arrow is the quotient by the ideal $\Pp$. It follows from Lemma \ref{goodinit} that we can now prove:
\begin{theorem}{\rm (Structure of rational valuations)}\label{structure} Let $R$ be a complete equicharacteristic noetherian local domain and let $k\subset R$ be a field of representatives of the residue field of $R$. Let $\nu$ be a rational valuation of $R$ and let $(\xi_i)_{i\in I}$ be representatives in $R$ of a minimal set of generators $(\overline \xi_i)_{i\in I}$ of the $k$-algebra ${\rm gr}_\nu R$ for which the valuative Cohen theorem holds. Let $J\subset I$ be a minimal set such that the $(\xi_i)_{i\in J}$ generate the maximal ideal of $R$ and let $(U^{m_\ell}-\lambda_\ell U^{n_\ell})_{\ell\in L}$ be a minimal set of generators of the kernel of the surjective map of $k$-algebras $k[(U_i)_{i\in I}]\to {\rm gr}_\nu R,\ \ U_i\mapsto\overline\xi_i$.\par
Then a set of topological generators for the kernel of the continuous surjective map $\pi\colon\widehat{k[(u_i)_{i\in I}]}\to R$ given by the valuative Cohen theorem is as follows: 
\begin{itemize}
\item For each $i\in I\setminus J$, a generator of the form $$F_i=u^{m^i}-\lambda_i u^{n^i}-g_i((u_j)_{j<i})-u_i,$$ where the initial binomial is one of the  $(U^{m_\ell}-\lambda_\ell U^{n_\ell})_{\ell\in L}$, the series $g_i((u_j)_{j<i})\in \widehat{k[(u_i)_{i\in I}]}$ depends only on variables of index $<i$ and the weight of each of its terms is $>w(u^{m^i})$ and $<w(u_i)$.
\item A finite subset of the complement in $L$ of the set of binomials used above is in bijection with generators of the form $$F_q=u^{m^q}-\lambda_q u^{n^q}-g_q(u),$$ with $g_q(u) \in \widehat{k[(u_i)_{i\in I}]}$ and the weight of each of its terms is $>w(u^{m^q})$.
\end{itemize}
Moreover, each of these generators uses only finitely many of the variables $u_i$.
\end{theorem}
\begin{proof}  From the diagram above we see that we have the equality $\Pi^{-1}(P)={\rm ker}\Pi+\Pp$ and that ${\rm ker}\pi$ is the image of this ideal in $\widehat{k[(u_i)_{i\in I}]}$. So we obtain topological generators of ${\rm ker}\pi$ by reducing the topological generators of ${\rm ker}\Pi$ modulo $\Pp$. By Lemma \ref{goodinit}, those which involve a $-w_a$ as a term vanish entirely,  those which involve a $-v_q$ as a last term do not vanish since their initial binomial involves only the $u_i$ and become the $F_q$ of the theorem. Those which involve a $-u_i$ as a last term pass to the quotient without modification and become the $F_i\in {\rm ker}\pi$.    
\end{proof}
\begin{remark}\label{str}1) One cannot expect that in general the initial binomials of the series $F_i, F_q$ are all the $(U^{m^\ell}-\lambda_\ell U^{n^\ell})_{\ell\in L}$. As in the regular case, there is no reason why these topological generators should be a (generalized) standard basis.\par\noindent
2) We know that, with the notations of Lemma \ref{upgen}, we can write each $p_q$ as a series in the $(\tilde\eta_j)_{j\in J}$. In fact the map $\Pi$ induces an isomorphism from the sub $k$-algebra $k[[(u_j)_{j\in J}]]$ of $\widehat{k[(u_i)_{i\in I},v_1,\ldots ,v_s,(w_a)_{a\in A}]}$ to $S$ and the $p_q((u_j)_{j\in J})-v_q$ belong to the kernel of $\Pi$. \textit{The point is to write it, as we do using the topological generator of ${\rm ker}\Pi$ which contains $-v_q$ as a term, using only elimination through equations whose initial forms are binomials belonging to our set of generators of the kernel of ${\rm gr}_w\Pi$. This is what will allow us to obtain a quotient of $R$ as an overweight deformation of a finite number of these binomials.}\par\noindent
3) The theorem shows the persistence of the abyssal phenomenon of Remark \ref{choice} even when $R$ is not regular.\par\noindent
4) An interpretation of the theorem is that it presents the valuation $\nu$ on $R$ as the image of the \textit{monomial} valuation on $\widehat{k[(u_i)_{i\in I}]}$ given by the weights, in the sense that the value of an element of $R$ is the maximum weight of its representatives in $\widehat{k[(u_i)_{i\in I}]}$ (see \cite[Proposition 3.3]{Te3}). \par\noindent
5) The semigroup $\Delta$ of values of the valuation $\mu$ on $S$ requires more generators than the $(0,\gamma_i)$, the $\mu(p_j)$ and the $(f,\gamma_i),\ f\in\N, f\geq 1$, and their description would be interesting. We do not need it here thanks to Lemma \ref{goodinit}.
\end{remark}
\end{subsection}
\begin{subsection}{Equations for the toric degeneration}
We can now write the equations for the toric degeneration of $R$ to ${\rm gr}_\nu R$ in the natural parameters, the $v^\varphi$, where $\varphi$ runs through a system of generators for the semigroup $\Phi_{\geq 0}$, although there is not a minimal one in general. See \cite[\S 4]{Te1}.\par
Recall from \cite[Section 2.3]{Te1} that the algebra encoding the toric degeneration of the ring $R$ to its associated graded ring ${\rm gr}_\nu R$ is the valuation algebra $$\Aa_\nu(R)=\bigoplus_{\varphi\in\Phi} \Pp_\varphi(R)v^{-\varphi}\subset R[v^\Phi].$$
Having fixed a field of representatives $k\subset R$ of the residue field, the composed injection $k[v^{\Phi_{\geq 0}}]\subset R[v^{\Phi_{\geq 0}}]\subset \Aa_\nu(R)$ stemming from the fact that $\Pp_\varphi(R)=R$ for $\varphi\leq 0$ gives $\Aa_\nu(R)$ the structure of a faithfully flat $k[v^{\Phi_{\geq 0}}]$-algebra and according to \cite[Proposition 2.2]{Te1}, the map $${\rm Spec}\Aa_\nu (R)\rightarrow {\rm Spec}k[v^{\Phi_{\geq 0}}]$$ has special fiber ${\rm Spec}{\rm gr}_\nu R$ and its general fiber is isomorphic to $R$ in the sense that, denoting by $(v^{\Phi_+})$ the multiplicative subset of $R[v^{\Phi_{\geq 0}}]$ consisting of elements $v^\varphi,\ \varphi>0$, the natural inclusion $R[v^{\Phi_{\geq 0}}]\subset \Aa_\nu (R)$ extends after localization to an isomorphism of $(v^{\Phi_+})^{-1}k[v^{\Phi_{\geq 0}}]$-algebras:
$$(v^{\Phi_+})^{-1}R[v^{\Phi_{\geq 0}}]\simeq (v^{\Phi_+})^{-1}\Aa_\nu(R).$$
Let us consider the $k[v^{\Phi_{\geq 0}}]$-algebra $k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}=k[v^{\Phi_{\geq 0}}]\otimes_k\widehat{k[(\tilde u_i)_{i\in I}]}$, which is the same construction as $\widehat{k[(u_i)_{i\in I}]}$ in section \ref{reminder} but where the series have coefficients in $k[v^{\Phi_{\geq 0}}]$. The variable $\tilde u_i$ still has weight $\gamma_i$. Remember from \cite[Section 2.4]{Te1} that we can define on the algebra $\Aa_\nu(R)$ a valuation $\nu_\Aa$ by $$\nu_\Aa(\sum x_\varphi v^{-\varphi})={\rm min}( \nu(x_\varphi)),\ {\rm with}\ {\rm gr}_{\nu_\Aa}\Aa_\nu (R)\simeq k[v^{\Phi_{\geq 0}}]\otimes_k {\rm gr}_\nu R.$$
\begin{lemma}\label{dyngen}
With the notations of Theorem \ref{cohen}, the $(\xi_iv^{-\gamma_i})_{i\in I}$ constitute a minimal system of homogeneous topological generators of the $k[v^{\Phi_{\geq 0}}]$-algebra $\Aa_\nu (R)$ in the sense that the map of $k[v^{\Phi_{\geq 0}}]$-algebras
$$\tilde \pi\colon k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}\longrightarrow \Aa_\nu(R),\ \ \tilde u_i\mapsto \xi_iv^{-\gamma_i}$$
is surjective and continuous for the topologies defined by the weight and the valuation $\nu_\Aa$ respectively.
\end{lemma}
\begin{proof} By the valuative Cohen theorem (Theorem \ref{cohen}) if $x\in\Pp_\varphi(R)$ we can write $x=\sum_e d_e\xi^e$ where the $d_e\in k^*$ and the $\xi^e$ are monomials in the $\xi_i$ of increasing valuations, with the minimum value being $\geq \varphi$. Let us write $-\varphi=-\nu(\xi^e)+\eta_e$ with $\eta_e\geq 0$. Note that the $\eta_e$ corresponding to the smallest index in the sum is zero if $\nu(x)=\varphi$ and all the others are $>0$. Then we can write $xv^{-\varphi}=\sum_ed_ev^{\eta_e}\xi^ev^{-\nu(\xi^e)}$ and since the $\xi^ev^{-\nu(\xi^e)}$ are monomials in the $\xi_iv^{-\gamma_i}$ and the minimality is clear as well as the continuity, this proves the result. \end{proof}
We now think of the $\tilde u_i=\xi_iv^{-\gamma_i}$ as defining an embedding of the total space of the toric degeneration into the space corresponding to the algebra $k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$. We are seeking equations for the image of this embedding, that is, generators for the kernel of the map $\tilde \pi$ of Lemma \ref{dyngen} above.\par\noindent
If a series $H((\tilde u_i))_{i\in I})$ is in the kernel, we see that by definition of the map, we have $H((v^{\gamma_i}\tilde u_i)_{i\in I})=0$ in $R$, and conversely. so that the change of variables $u_i\mapsto v^{\gamma_i}\tilde u_i$, which is well defined over $(v^{\Phi_+})^{-1}k[v^{\Phi_{\geq 0}}]$, carries the kernel of the map $\pi$ of Theorem \ref{cohen} to the kernel of $\tilde\pi$. It realizes the isomorphism, valid over $(v^{\Phi_+})^{-1}k[v^{\Phi_{\geq 0}}]$  of the fiber of ${\rm Spec}\Aa_\nu (R)\to {\rm Spec}k[v^{\Phi_{\geq 0}}]$ over a point where "all the $v^\varphi,\ \varphi>0$ are $\neq 0$" with the space corresponding to $R$.\par\noindent  Now for each topological generator $F_\ell=u^{m^\ell}-\lambda_\ell u^{n^\ell}+\cdots$ of the kernel of $\pi$ we see that $F_\ell (v^\gamma \tilde u)$, with the obvious contracted notation, is divisible, as an element of $k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$, by $v^{w(m^\ell)}=v^{w(n^\ell)}$. The ideal of $k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$ topologically generated by the $$\tilde F_\ell=v^{-w(m^\ell)}F_\ell(v^\gamma\tilde u)=\tilde u^{m^\ell}-\lambda_\ell \tilde u^{n^\ell}+\cdots$$ is contained in the kernel of $\tilde\pi$, is equal to it and also isomorphic to the kernel of $\pi$ over $(v^{\Phi_+})^{-1}k[v^{\Phi_{\geq 0}}]$, while the space it defines modulo the ideal $(v^{\Phi_{>0}})$ is the same as the space defined by $\Aa_\nu(R)$ modulo that ideal, namely ${\rm Spec}{\rm gr}_\nu R$. By flatness of $\Aa_\nu(R)$ over $k[v^{\Phi_{\geq 0}}]$, they have to coincide.\par\medskip
Combining this with Theorem \ref{structure} we have obtained the following:
\begin{theorem}
With the notations of Theorem \ref{structure}, a set of topological generators in $k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$ for the ideal defining the total space of the degeneration of $R$ to ${\rm gr}_\nu R$ is as follows:
\begin{itemize}
\item For each $i\in I\setminus J$, a generator of the form $$\tilde F_i=\tilde u^{m^i}-\lambda_i \tilde u^{n^i}-\tilde g_i(v,(\tilde u_j)_{j<i})-v^{\gamma_i-w(\tilde u^{m^i)}}\tilde u_i,$$ where the initial binomial is one of the  $(U^{m_\ell}-\lambda_\ell U^{n_\ell})_{\ell\in L}$, the series $\tilde g_i(v,(\tilde u_j)_{j<i})\in k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$ depends only on variables of index $<i$ and the weight of each of its terms is $>w(\tilde u^{m^i})$ and $<w(\tilde u_i)$.
\item A finite subset of the complement in $L$ of the set of binomials \break $(\tilde u^{m_\ell}-\lambda_\ell \tilde u^{n_\ell})_{\ell\in L}$ used by the $\tilde F_i$ is in bijection with generators of the form $$\tilde F_q=\tilde u^{m^q}-\lambda_q \tilde u^{n^q}-\tilde g_q(v,\tilde u),$$ with $\tilde g_q(v,\tilde u) \in k[v^{\Phi_{\geq 0}}]\widehat{[(\tilde u_i)_{i\in I}]}$ and the weight of each of its terms is $>w(\tilde u^{m^q})$.
\end{itemize}
Each series uses finitely many of the variables $\tilde u_i$ and each term appearing in the series $\tilde F_i$ or $\tilde F_q$ is of the form $c_ev^{w(\tilde u^e)-w(\tilde u^{m^\ell})}\tilde u^e$ or $c_ev^{w(\tilde u^e)-w(\tilde u^{m^q})}\tilde u^e$ with $c_e\in k^*$.
\end{theorem}
\begin{remark}  In view of  \cite[Proposition 2.3]{Te3}, the ring $k[v^{\Phi_{\geq 0}}]$ is the union of a nested sequence of polynomial rings in $r$ variables, where $r$ is the rational rank of $\Phi$:
\small{$$k[x_1^{(0)},\ldots ,x_r^{(0)}]\subset\ldots \subset k[x_1^{(h)},\ldots ,x_r^{(h)}]\subset k[x_1^{(h+1)},\ldots ,x_r^{(h+1)}]\subset\ldots\subset k[v^{\Phi_{\geq 0}}],$$}\rm
where each variable $x_i^{(h)}$ is mapped to a monomial in the $x_j^{(h+1)},\ j=1,\ldots ,r$. Therefore whenever we deal with a finite set of series $F_\ell$, for example the $F_\ell\vert B$ of section \ref{AbhApp} below, since we may assume that they use only finitely many of the variables $u_i$, the parameter space of the toric degeneration can be taken to be an $r$-dimensional affine space.
\end{remark}
\begin{example}\label{ex2}
 Consider the overweight deformation, as in section \ref{ow}: $$y^2-x^3-u, \ u^2-x^{s-1}y$$ of the binomial ideal $y^2-x^3, u^2-x^{s-1}y$, where $x,y,u$ have weights $4,6,2s+1$ respectively, with $s\geq 6$. Set $\tilde x=v^{-4}x,\tilde y=v^{-6}y, \tilde u=v^{-(2s+1)}u$. \par\noindent Substituting $x=v^4\tilde x,y=v^6\tilde y,u=v^{2s+1}\tilde u$ in the equations, we obtain $v^{12}\tilde y^2-v^{12}\tilde x^3-v^{2s+1}\tilde u=0, v^{4s+2}\tilde u^2-v^{4s+2}\tilde x^{s-1}\tilde y=0$. \par After dividing each equation by the highest possible power of $v$, we obtain the equations for the total space of the toric degeneration over ${\rm Spec}k[v]$ of the formal curve in $\A^3(k)$ with ring $R=k[[x,y,u]]/(y^2-x^3-u,u^2-x^{s-1}y)$ to the monomial (toric) curve $\tilde x=t^4, \tilde y=t^6, \tilde u=t^{2s+1}$: $$\tilde y^2-\tilde x^3-v^{2(s-6)+1}\tilde u=0, \ \tilde u^2-\tilde x^{s-1}\tilde y=0.$$
 \end{example}
 \begin{example} Let us go back to Example \ref{ex1} and let $R\subset k[[t]]$ be the algebras of a plane branch and its normalization, endowed with the $t$-adic valuation. We have an inclusion of valuation algebras $\Aa_\nu (R)\subset \Aa_\nu (k[[t]])\simeq k[v][[t]][v^{-1}t].$ As above, we introduce $\tilde t=v^{-1}t$ to embed the space corresponding to $\Aa_\nu (k[[t]])$ in the affine space $\A^1(k[v])$ over the affine line. The $\xi_i\in R$ of Example \ref{ex1}, viewed as elements $\xi_i(t)\in k[[t]]$, parametrize the embedding of our curve in $\A_k^{g+1}$. They can be written
 $$\xi_i(t)=t^{\gamma_i}+\sum_{e>\gamma_i}c_e^{(i)}t^e,\ \ {\rm with}\ c_e^{(i)}\in k.$$
  The generators $v^{-\gamma_i}\xi_i$ of $\Aa_\nu (R)$ then become elements $v^{-\gamma_i}\xi_i(v\tilde t)\in k[v][[\tilde t]]$ which realize in a parametric way the degeneration of our curve to the monomial curve with the same semigroup, as in \cite[Section 3]{G-T}. 
  $$v^{-\gamma_i}\xi_i(v\tilde t)=\tilde \xi_i(v,\tilde t)=\tilde t^{\gamma_i}+\sum_{e>\gamma_i}c_e^{(i)}v^{e-\gamma_i}\tilde t^e,\ \ {\rm with}\ c_e^{(i)}\in k.$$
  A version of this example appears in \cite[Section 5.4]{Te1}.
  \end{example}
One can compare the degenerations, by equations and by parametrization, on the following example. It is the case $s=6$ of Example \ref{ex2}, where the equation of the plane branch has been changed a little in order to simplify the parametrization, and we assume characteristic $\neq 2$.\par\noindent The equations and the parametrization correspond to the "torific" embedding in $\A^3(k)$ of the plane branch with equation $(y^2-x^3)^2-x^5y-\frac{1}{16}x^7=0$.
$$\tilde y^2-\tilde x^3-v\tilde u=0,\ \ \tilde u^2-\tilde x^5\tilde y-\frac{1}{16}v^2\tilde x^7=0$$
$$\tilde x=\tilde t^4, \tilde y=\tilde t^6+\frac{1}{2}v\tilde t^7, \tilde u=\tilde t^{13}+\frac{1}{4}v\tilde t^{14}.$$
\begin{remark}\label{homog}Since $\Aa_\nu(R)$ is a $\Phi$-graded algebra, the component of degree $\varphi$ being $\Pp_\varphi v^{-\varphi}$, it is natural that equations and parametrizations for the toric degeneration should be homogeneous (or rather isobaric) when $v$ is given weight $-1$.
\end{remark}
\end{subsection}

\end{section}
\begin{section}{\small{The basic example of approximation of a valuation by Abhyankar semivaluations}}\label{ex}

Let $k$ be an algebraically closed field and let $\nu$ be a rational valuation of rational rank one on the ring $R=k[[x,y]]$. It cannot be Abhyankar, and therefore the semigroup $\Gamma=\nu (R\setminus\{0\})$ is not finitely generated (see \cite[beginning of \S 7]{Te3}). Up to a change of variables we may assume that $\nu(x)<\nu(y)$ and that $\nu(y)$
is the smallest non zero element of $\Gamma$ which is not in $\Gamma_0=\N\nu(x)$. By \cite[Theorem 1.1 and Lemma 2.1]{C-V}, the semigroup $\Gamma$ is generated by positive rational numbers $\nu(x),\nu(y),(\gamma_i)_{i\geq 2}$ and we can normalize the valuation by setting $\gamma_0=\nu(x)=1$. Here, as in Example \ref{ex1}, we denote the smallest element of $\Gamma$ by $\gamma_0$ instead of $\gamma_1$ to underscore the special role played historically by the coordinate $x$.\par\noindent Moreover, denoting by $\Phi_i$ (resp. $\Gamma_i$) the subgroup (resp. subsemigroup)  of $\Q$ generated by $1,\nu(y),\nu(u_2),\ldots ,\nu(u_i)$, with $\Phi_0=\Z$, $\Phi_1$ (resp. $\Gamma_1$) generated by $\nu(x),\nu(y)$, and denoting by $n_i$ the index $[\Phi_i:\Phi_{i-1}]$, we have that $\gamma_{i+1}>n_i\gamma_i$ and as a consequence $n_i\gamma_i\in \Gamma_{i-1}$ and the relations between the $\gamma_i$ are generated by the expressions of this last result as $$n_i\gamma_i= t^{(i)}_0\gamma_0+t^{(i)}_1\gamma_1+\sum_{j=2}^{i-1}t^{(i)}_j\gamma_j\ \ {\rm with}\ t^{(i)}_j\in\N,\ \ 0\leq j\leq i-1,$$
where we may assume that $t^{(i)}_j<n_j$ for $j\geq 1$.\par\noindent
According to Theorem \ref{regular}, by a good choice of the representatives $\xi_i\in k[[x,y]]$ of the generators $\gamma_i$ the map 
\[\widehat{k[x,y,(u_i)_{i\geq 2}]}\to k[[x,y]],\ x\mapsto x, y\mapsto y, u_i\mapsto \xi_i,\] is well defined and surjective, and
has a kernel generated up to closure by $$\begin{array}{lr}y^{n_1}-x^{t^{(1)}_0}-g_1(x,y)-u_3,\\ \vdots \\ u_i^{n_i}-x^{t^{(i)}_0}y^{t^{(i)}_1}\prod_{2\leq q\leq i-1} u_q^{t^{(i)}_q}-g_i(x,y,(u_j)_{2\leq j\leq i})-u_{i+1}\\
u_{i+1}^{n_{i+1}}-x^{t^{(i+1)}_0}y^{t^{(i+1)}_1}\prod_{2\leq j\leq i } u_j^{t^{(i+1)}_j}-g_{i+1}(x,y,(u_j)_{2\leq j\leq i+1})-u_{i+2}
\\ \vdots\\ \vdots \end{array}$$ with $g_i(x,y,(u_j)_{2\leq j\leq i})\in k[[(x,y,(u_j)_{2\leq j\leq i}]]$ satisfying the overweight condition and having each term of weight $<\gamma_{i+1}$.\par
Now if we keep only the first $i$ equations, and set $u_j=0$ for $j\geq i+1$, we obtain by elimination of the variables $u_j$ , $2\leq j\leq i$ the equation $p_i(x,y)=0$ of a plane branch $C_i$ in the formal affine plane. It has a unique valuation, which is Abhyankar and has a semigroup $\Gamma_i$ generated by $1,\nu(y),\nu(u_j)_{3\leq j\leq i}$. The unique valuation on the branch $C_i$ is an Abhyankar semivaluation on $k[[x,y]]$ and the corresponding semigroups $\Gamma_i=\langle 1,\gamma_1,\gamma_2,\ldots ,\gamma_i\rangle$ fill the semigroup $\Gamma$ as $i$ grows. The intersection numbers with $C_i$ at the origin of the elements $\xi_j\in k[[x,y]]$ are the $\gamma_j$, for $2\leq j\leq i$. Finally, by Chevalley's theorem, there exists a function $\beta\colon\N\to \N$ with $\beta(i)$ tending to infinity with $i$, such that $p_i(x,y)\in (x,y)^{\beta(i)}$.\par
To the branch $C_i$ is associated a rank two valuation $\nu_i$ on $k[[x,y]]$ defined as follows: each element $h\in k[[x,y]]$ can be written uniquely as $h=p_i^n\tilde h$ with $\tilde h$ not divisible by $p_i$ and then the restriction of $\tilde h$ to $C_i$ has a value $\gamma\in \Gamma_i$.\par\noindent The map $h\mapsto (n,\gamma)\in \Z\times \Phi_i$ 
is the rank two valuation $\nu_i$, which is a rational Abhyankar valuation dominating $k[[x,y]]$. The valuations $\nu_i$ approximate $\nu$ in the sense that given any $h\in k[[x,y]]$, for sufficiently large $i$ the element  $p_i(x,y)\in (x,y)^{\beta(i)}$ does not divide $h$ so that $\nu_i(h)\in \{0\}\times \Gamma_i$ is also $\nu(h)$ by what we have seen above. So we can indeed approximate the non-Abhyankar valuation $\nu$ of rank one and rational rank one by Abhyankar valuations of $k[[x,y]]$ of rank two and rational rank two, but the fundamental fact is the approximation of $\nu$ by Abhyankar semivaluations.
\begin{remark} If the algebraically closed field $k$ is of characteristic zero, each equation $p_i(x,y)=0$ has for roots Puiseux expansions with $i+1$ characteristic exponents $\beta_0,\ldots,\beta_i$ which coincide, up to renormalization, with Puiseux expansions up to and excluding the $(i+1)$-th Puiseux exponents of the roots of the equations $p_j=0$ for $j>i$ (see \cite{PP}). As $i$ tends to $\infty$ these Puiseux series converge to a series $y_\infty(x)\in k[[x^{\Q_{\geq 0}}]]$ which is not a root of any polynomial or power series in $x,y$ because the denominators of the exponents are not bounded. Given $h(x,y)\in k[[x,y]]$, associating to it the order in $x$ of $h(x,y_\infty(x))$ defines a valuation, which is the valuation $\nu$. For more details in a more general situation, see \cite{M}.\par\noindent The relationship between these Puiseux exponents $\beta_j$ and the generators of the semigroup is quite simple (see \cite[Chap.1]{Z2}): it is $\gamma_{j+1}-n_j\gamma_j=\beta_{j+1}-\beta_j$.\par These beautiful facts are unfortunately missing when one tries to understand valuations over fields of positive characteristic.  Even for branches the behaviour of solutions in generalized power series $y(x)\in k[[x^{\Q_{\geq 0}}]]$ of algebraic equations is much more complicated when $k$ is of positive characteristic (see \cite{K}) and as far as we know the relationship of these solutions with the semigroup of values has not been clarified (see \cite[Section 1]{Te4}).\end{remark}


\end{section}
\begin{section}{Approximating a valuation by Abhyankar semivaluations}\label{AbhApp}
The purpose of this section is to show that the situation described in Example \ref{ex} is in fact quite general. We start from Theorem \ref{structure}. Let us choose a finite initial subset $B_0$ of $I$ which contains:\begin{itemize}
\item The set $J$ of indices of the elements $(\xi_i)_{i\in J}$ minimally generating the maximal ideal of $R$;
\item A set of $r={\rm rat.rk.}\Phi$ indices $i_1,\ldots ,i_r$ such that the $(\gamma_{i_t})_{t=1,\ldots ,r}$ rationally generate the group $\Phi$;
\item The indices of the finite set of variables $u_i$ appearing in the equations $(F_q)_{q=1,\ldots ,s}$ of Theorem \ref{structure}.
\end{itemize}
Let $B$ be a finite initial subset of $I$ containing $B_0$ and consider in $R$ the ideal $K_B$ generated by the $(\xi_i)_{i\notin B}$ and the images $F_\ell\vert B$ by the map $\pi$ of the $F_\ell$ whose initial binomial involves only variables whose index is in $B$ and in which all the $u_i$ with $i\notin B$ have been set equal to $0$. Note that only the $F_\ell$ which use only variables in $B$ are mapped to zero in this operation. 
\begin{proposition}\label{quot}
\begin{enumerate}
\item The quotient $R/K_B$ is an overweight deformation of the prime binomial ideal generated by the binomials $u^{m^\ell}-\lambda_\ell u^{n^\ell}$ which are contained in $k[(u_i)_{i\in B}]$. In particular, $K_B$ is a prime ideal.
\item Given an integer $D$ there exists a finite initial set $B(D)$ such that $K_{B(D)}\subset m^{D+1}$.
\end{enumerate}
\end{proposition}
\begin{proof}
The following result is classical, see \cite[Lemma 1-90]{G-L-S} for the complex analytic avatar:

 \begin{lemma}\label{dim} Let $R_0\subset R$ be an injection of complete local noetherian $k$-algebras with residue field $k$. Then we have ${\rm dim}R\geq {\rm dim}R_0$.
 \end{lemma}
 By the formal normalization lemma, there exist elements $x_1,\ldots ,x_e$ in $R\setminus m_0R$ such that the induced map of $k$-algebras $k[[X_1,\ldots ,X_e]] \to R/m_0R$ sending $X_i$ to the image of $x_i$ is injective and finite.\par\noindent By the Weierstrass preparation theorem in the form "quasi-finite implies finite", the map $R_0[[X_1,\ldots ,X_e]]\to R$ sending $X_i$ to $x_i$ is also injective and finite, so that ${\rm dim} R={\rm dim} R_0+e\geq {\rm dim} R_0$. \par\medskip\noindent

 \textit{In the following lemma, an} overweight unfolding \textit{is a deformations of equations which adds only terms of higher weight, without any condition on the initial forms of the elements of the ideal generated by the deformed equations.}
 \begin{lemma}\label{flat} Let $F_0=(u^{m^\ell}-\lambda_\ell u^{n^\ell})_{\ell \in L}$ be a prime binomial ideal in $k[[u_1,\ldots ,u_N]]$ corresponding to a system of generators of the relations between the generators of a finitely generated semigroup $\Gamma=\langle \gamma_1,\ldots ,\gamma_N\rangle$ generating a group of rational rank $r$. Let $w$ be the weight  on $k[[u_1,\ldots ,u_N]]$ defined by giving $u_i$ the weight $\gamma_i$.\par Let $F_\ell=u^{m^\ell}-\lambda_\ell u^{n^\ell} +\sum_{w(u^p)>w(u^{m^\ell})}c^{(\ell)}_pu^p)_{\ell\in L}$ be overweight unfoldings of the binomials. Let $F$ be the ideal of $k[[u_1,\ldots ,u_N]]$ generated by the $(F_\ell)_{\ell\in L}$ and $R=k[[u_1,\ldots ,u_N]]/F$.\par\noindent If ${\rm dim}R\geq r$, then ${\rm dim}R= r$ and the unfolding is an overweight deformation of the binomial ideal.
 \end{lemma}
 Exactly as in the proof of Theorem 3.3 of \cite{Te3}, we can define an order function with values in $\Gamma$ on $R$ by associating to a non zero element of $R$ the highest weight of its preimages in $k[[u_1,\ldots ,u_N]]$. Then we have, as in \cite[subsection 2.3]{Te1},
 a faithfully flat specialization of $R$ to $k[[u_1,\ldots ,u_N]]/{\rm in}_wF$, where ${\rm in}_wF$ is the $w$-initial ideal of the ideal $F$, and ${\rm dim}(k[[u_1,\ldots ,u_N]]/{\rm in}_wF)={\rm dim}R$. On the other hand we have the exact sequence
 \[(0)\rightarrow {\rm in}_w F/F_0\rightarrow k[[u_1,\ldots ,u_N]]/F_0\rightarrow k[[u_1,\ldots ,u_N]]/{\rm in}_wF\rightarrow(0),\]
 and since the domain $ k[[u_1,\ldots ,u_N]]/F_0$ is of dimension $r$ (see \cite[Proposition 3.7]{Te1}), the inequality ${\rm dim}R\geq r$ is equivalent to the inequality \[{\rm dim}k[[u_1,\ldots ,u_N]]/{\rm in}_wF\geq {\rm dim}k[u_1,\ldots ,u_N]/F_0\] which implies the equality of dimensions and, since $F_0$ is prime, the equality ${\rm in}_wF=F_0$ which means that we have an overweight deformation.

  By construction, the ring $R/K_B$ is a quotient of $k[[(u_i)_{i\in B}]]$ by the ideal generated by the $ F_\ell\vert B$ which are the $F_\ell$ whose initial forms are in $k[(u_i)_{i\in B}]$ and where we have set $u_i=0$ if $i\notin B$. So it is an overweight unfolding of the ideal generated by its initial binomials. The quotient of $k[(u_i)_{i\in B}]$ by this binomial ideal is isomorphic by Proposition \ref{smgrp} to the semigroup algebra $k[t^{\Gamma_i}]$ where $\Gamma_i$ is the semigroup generated by the $(\gamma_i)_{i\in B}$ and so has dimension $r$.\par  By Lemmas \ref{dim} and \ref{flat}, to prove the first assertion of the proposition it suffices to prove that $R/K_B$ contains the power series ring $k[[u_{i_1},\ldots ,u_{i_r}]]$. But $(i_1,\ldots ,i_r)\subset B$ by our choice of $B_0$ and a series $h(u_{i_1},\ldots ,u_{i_r}) \in k[(u_i)_{i\in I}]$ cannot belong to the ideal of the $F_\ell$ used to define $K_B$: since the weight of the variables are rationally independant the initial form of $h(u_{i_1},\ldots ,u_{i_r})$ is a monomial and the initial ideal of the $F_\ell$ contains no monomial. \par\noindent Thus, the map $k[[u_{i_1},\ldots ,u_{i_r}]]\to R/K_B$ is injective so that ${\rm dim}R/K_B\geq r$ by Lemma \ref{dim} and we have the result by Lemma \ref{flat},\par\noindent
The second part of the theorem follows from the fact that by Theorem \ref{Chevalley} we can choose $B=B(D)$ so that the $(\xi_i)_{i\notin B}$ are all in $m^{D+1}$ and by construction the image in $R$ of $ F_\ell\vert B\in k[[(u_i)_{i\in B}]]$ differs from zero by a series all of whose terms are in the ideal generated by the $(\xi_i)_{i\notin B}$.
   \end{proof} 
To summarize, the ring  $R/K_B$  is indeed an overweight deformation of the prime binomial generated by the $u^{m^\ell}-\lambda_\ell u^{n^\ell}$ which are in $k[[(u_i)_{i\in B}]]$ in the sense of  subsection \ref{ow} and \cite[Section 3]{Te3}, and the induced valuation has to be Abhyankar, of rational rank $r={\rm rat.rk.}\nu={\rm dim}R/K_B$. The value semigroup of this Abhyankar valuation, which we shall denote by $\nu_B$, is the subsemigroup of $\Gamma$ generated by the $(\gamma_i)_{i\in B}$.\par\noindent
  We now study some properties of the collection of ideals $K_{B}$.\par\medskip
 Notice that given an inclusion $B\subset B'$, the set of generators of the ideal $K_{B'}$ contains less of the variables $u_i$, but in general more of the images of the series $F_\ell$ because the variables with indices in $B'\setminus B$ will in general appear in the initial forms of some new $F_\ell$'s. So we have:
 \begin{remark}\label{order}In general when  $B\subset B'$ there is no inclusion between the ideals $K_B$  and $K_{B'}$. However, it follows from Theorem \ref{Chevalley} that if for some integer $D$ we have $K_B\subset m^{D+1}$ and $B\subset B'$, then $K_{B'}\subset m^{D+1}$. The $m$-adic order of the ideals $K_B$ grows with $B$ and tends to infinity.
 \end{remark}
 \begin{corollary}\label{vanish} With the notations of Corollary \ref{approx}, we have $\bigcap_{B\supset B_0}K_{B}=\bigcap_{t\in\N}K_{B_t}=(0)$.
 \end{corollary}
 \begin{proof}This follows from the fact that $R$ is noetherian and the second part of the proposition. \end{proof}
 \begin{corollary}\label{jacobian}
 Let $J\subset R$ be the jacobian ideal, defining the singular locus of the formal germ associated to $R$.  There exists a finite initial set $B_J$ containing $B_0$ and such that if an initial set $B$ contains $B_J$ we have that $J\not\subset K_{B}$ so that the localization $R_{K_{B}}$ is a regular ring. In particular, in the sequence of Corollary \ref{approx}, we have that $J\not\subset K_{B_t}$ and $R_{K_{B_t}}$ is a regular ring for large enough $t$.
 \end{corollary}
 \begin{proof} Since $R$ is a domain and a quotient of a power series ring, we know that the ideal $J$ is not zero. Since $R$ is noetherian, we know that $\bigcap_{j=1}^\infty m^j=(0)$. Let $D$ be the largest integer $E$ such that $J\subset m^E$ in $R$. If we take $B_J=B(D)$ as in the proposition, we have that $K_{B_J}\subset m^{D+1}$. Thus, if $B_J\subset B$ it is impossible for $J$ to be contained in $K_{B}$.\end{proof}
 
\begin{theorem}\label{AbhApprox} The Abhyankar semivaluations $\nu_B$ of $R$ are better and better approximations of the valuation $\nu$ as the finite initial sets $B$ grow in the sense that:
\begin{enumerate} 
\item For any $x\in R\setminus\{0\}$, let $D$ be its $m$-adic order. By Theorem \ref{Chevalley} there are finite initial sets $B$ containing $B_0$ such that $K_B\subset m^{D+1}$. Then for any finite initial set $B'\supset B$ we have $x\notin K_{B'}$ and for any such $B'$ we have the equality $\nu(x)=\nu_{B'}(x)$. In particular, for any $x\in R\setminus\{0\}$ there is an index $t_0(x)$ in the sequence of Corollary \ref{approx} such that for $t\geq t_0(x)$ we have $x\notin K_{B_t}$ and $\nu(x)=\nu_{B_t}(x)$.
\item The union of the finitely generated semigroups $\Gamma_B=\langle(\gamma_i)_{i\in B}\rangle$ is equal to $\Gamma$, and in particular the nested union of the $\Gamma_{B_t}$ corresponding to the sequence of Corollary \ref{approx} is equal to $\Gamma$.
\end{enumerate}
\end{theorem}
\begin{proof} In view of Corollaries \ref{approx} and \ref{vanish} and of Remark \ref{order} it suffices to show that if $x\in R\setminus K_B$ its $\nu$-value is equal to its $\nu_B$-value. Consider the following diagram, which results from our constructions. 
\[\xymatrix{&k[[(u_i)_{i\in B}]]\ar[drr]_{\pi_B} \ar @{^{(}->}[r]^{\iota_B}&\widehat{k[(u_i)_{i\in I}]} \ar[r]^\pi&  R\ar[d]\\
              & && R/K_B}\leqno{(**)}\]
  The kernel of $\pi_B$ is generated by the $F_\ell\vert B$ as we have seen above. By definition (see \cite[Proposition 3.3]{Te3}) of the Abhyankar valuation on $R/K_B$, the $\nu_B$-value of an element $\overline x\in R/K_B$ is the maximum weight of its representatives in $k[[(u_i)_{i\in B}]]$. Let $h(\overline x)$ be such a representative. If its weight is not the $\nu$-value of its image in $R$ by the map $\pi$, its initial form must be in the binomial ideal generated by the initial forms of the $F_\ell$ (see the sentence after the statement of Theorem \ref{cohen} or \cite[Proposition 3.6]{Te3}). But then it is in the ideal generated by the initial forms of the $F_\ell\vert B$ and we have a contradiction.                  
  \end{proof}
\begin{remark}Just as in Example \ref{ex}, using Corollary \ref{jacobian}, we can, for large enough $B$, compose the valuation $\nu_B$ with the $K_BR_{K_B}$-adic valuation of the regular local ring $R_{K_B}$, to obtain a valuation $\tilde\nu_B$ on $R$ of rational rank $r+1\leq {\rm dim}R$. These valuations are Abhyankar if  $r+1={\rm dim}R$ and approximate $\nu$ in the sense that given $x\in R$, for large enough $B$ we have $\tilde\nu_B (x)=\nu(x)$ since $x\notin K_B$.
\end{remark}
\end{section}

\printindex
\end{document}
